NIST Special Publication 1277

A Framework to Evaluate the Cost-Effectiveness of Recovery-Based Design

Juan F. Fung Yating Zhang Katherine J. Johnson Dustin Cook Siamak Sattar

This publication is available free of charge from: https://doi.org/10.6028/NIST.SP.1277



NIST Special Publication 1277

A Framework to Evaluate the Cost-Effectiveness of Recovery-Based Design

Juan F. Fung Yating Zhang Applied Economics Office Engineering Laboratory

Katherine J. Johnson Dustin Cook Siamak Sattar Earthquake Engineering Group Engineering Laboratory

This publication is available free of charge from: https://doi.org/10.6028/NIST.SP.1277

May 2022



U.S. Department of Commerce Gina M. Raimondo, Secretary

National Institute of Standards and Technology Laurie E. Locasio, Director and Under Secretary of Commerce for Standards and Technology Certain commercial entities, equipment, or materials may be identified in this document in order to describe an experimental procedure or concept adequately. Such identification is not intended to imply recommendation or endorsement by the National Institute of Standards and Technology, nor is it intended to imply that the entities, materials, or equipment are necessarily the best available for the purpose.

National Institute of Standards and Technology Special Publication 1277 Natl. Inst. Stand. Technol. Spec. Publ. 1277, 81 pages (May 2022) CODEN: NSPUE2

> This publication is available free of charge from: https://doi.org/10.6028/NIST.SP.1277

Abstract

Modern building codes and standards are generally focused on safeguarding public health and safety by ensuring buildings meet life safety performance objectives following an earthquake event. Certain classes of buildings, such as hospitals and schools, may be designed to a higher standard to minimize disruption. As communities target resilience of their built environment, it is becoming evident that life safety performance can nevertheless result in massive disruptions for large classes of residential and commercial buildings. It is possible to design buildings to provide more rapid restoration of reoccupancy and functionality than life safety standards. Recoverybased design is intended to support a community's post-earthquake resilience and represents an advancement in design practice. This report presents a framework to provide guidance on the economic benefits and costs of adopting functional recovery in design practice. A unique aspect of the framework is the acknowledgement that benefits from recovery-based design are distributed across various community stakeholders (e.g., residents, customers, local government), in addition to stakeholders at the building level (e.g., owners and tenants). Thus, the framework carefully details the attribution of impacts to the range of stakeholders potentially affected by improved building-design criteria. A key objective of the framework is to identify the set of potential benefits and costs, impacted stakeholders, and available data and tools for quantification. In the process, we identify gaps in measurement science for conducting a complete benefit-cost analysis. In addition, we provide a brief case study example that illustrates an application of the framework.

Key words

Buildings, Community resilience, Earthquake, Economics, Functionality, Functional recovery, Recovery-based design, Reoccupancy

Acknowledgements

We are grateful to the Engineering Laboratory for FY21 Exploratory Project funding to support this research, as well as to Curt Haselton, Ed Almeter, and Dave Welch for their collaboration in developing the case study. In addition, we would like to extend our gratitude to Jennifer Helgeson (NIST), Emina Herovic (NIST), David Butry (NIST), and Nicos Martys (NIST) for their review, and to David Webb (NIST), Jennifer Helgeson (NIST), Anand Kandaswamy, and Joshua Kneifel (NIST) for their feedback. All errors are our own.

Table of Contents

Abstracti	
Acknowledgementsi	
Glossary iv	
1. Introduction and Motivation 1	
2. Background on Functional Recovery	
2.1. Development of seismic building codes and standards	
2.2. Performance-Based Earthquake Engineering 4	
2.3. What is functional recovery?	
3. Options for the Economic Evaluation of Recovery-Based Design	
3.1. Benefit-cost analysis	
3.1.1. Extensions of "traditional" BCA methods	.9
3.2. Cost-effectiveness analysis	
3.3. Life-cycle cost analysis	
3.4. Willingness-to-pay 11	
3.5. Economic impact analysis	
3.6. Summary	
4. Benefit-Cost Analysis for Building Design: A Review	
4.1. Benefits and costs of adopting higher standards for new construction	
4.2. Benefits and costs of seismically retrofitting existing buildings	
4.3. Benefits and costs of enhanced building designs in other domains	
5. The FR-BCA Framework	
5.1. Estimating benefits associated with enhanced building design	
5.1.1. Direct benefits	27
5.1.2. Indirect economic benefits	27
5.1.3. Insurance-related losses	28
5.1.4. Indirect property losses	29
5.1.5. Supply-chain disruption	30
5.1.6. Indirect social losses	31
5.1.7. Indirect physical losses	36
5.2. Estimating costs associated with enhanced building design	
5.3. Estimating building downtime	
5.3.1. Hazus	17
5.3.2. FEMA P-58	18
5.3.3. REDi and REDi-based methods	49
5.3.4. ATC-138 and fault tree-based methods	50
5.4. Case study	
6. Conclusion	

6.1. Summary/practical guidance	
6.2. Major findings	
6.3. Future directions	
References	
Appendix A. Risk category and seismic design category	
Appendix B. Methods/Tools for estimating costs	

List of Tables

Table 3.1. Alternative metrics for building investment decisions and their use for different types
of decisions. Adapted from ASTM E1185
Table 4.1. Benefit-transfer matrix (NIBS council 2019). 16
Table 5.1. Potential direct and indirect loss categories and the associated benefits from enhanced
design, as well as the potential beneficiaries and methods for monetizing
Table 5.2. Social cost of carbon from 2010 to 2050, in 2007 dollars per metric ton of CO ₂ .
Adapted from IWG (2016)
Table 5.3. Potential adoption/implementation costs, who bears cost, and how to estimate cost 40
Table 5.4. Assessment methods for downtime/functional recovery time estimation
Table 5.5. Software programs for downtime/functional recovery time estimation
Table 5.6. Case study benefit-cost analysis results (relative to baseline design). Low: $\delta = 0.07$,
$T = 50$. High: $\delta = 0.02, T = 75$
Table 5.7. Distributed benefit-cost analysis for archetype NS, based on the benefit-transfer
matrix in Table 4.1 (NIBS 2019). Weights for benefits are unnormalized
Table A-1. Risk category of buildings and other structures for earthquake loads (ASCE 7-16). 71
Table A-2. Seismic design category (ASCE 7-16). 73

List of Figures

Figure 1. Theoretical range of building performance and relative placement of safety-based and	nd
recovery-based goals. Source: FEMA P-2090/NIST SP1254	5
Figure 2. Structural cost estimating procedure.	42

Glossary

ABM: Agent-Based Model
BCA: Benefit-cost analysis
BCR: Benefit-cost ratio
CEA: Cost-effectiveness analysis
EAL: Expected Annualized Losses
FEMA: Federal Emergency Management Agency
FR: Functional recovery
FR-BCA: Functional recovery Benefit-cost analysis
IBC: International Building Code
ICC: International Code Council
I-O: Input-Output
IO: Immediate occupancy
IRR: Internal rate of return
LS: Life safety
NPV: Net present value
PBD: Performance-Based Design
PBEE: Performance-Based Earthquake Engineering
ROI: Return on investment
WTP: Willingness-to-pay
PTSD: Post-Traumatic Stress Disorder
QoL: Quality of Life
GDP: Gross Domestic Product
US&R: Urban Search and Rescue
VSL: Value of a Statistical Life

1. Introduction and Motivation

Recovery-based design has attracted attention from policy makers to the public, from the local to the national level, and represents the future of earthquake engineering (ICC 2019). Functional recovery is a "a post-earthquake performance state in which a building is maintained, or restored, to safely and adequately support the basic intended functions associated with the pre-earthquake use or occupancy of the building" (NIST 2021). Designing for functional recovery would be a notable shift in design philosophy from current safety-based objectives to recovery-based objectives. Identifying and defining recovery-based objectives would require the combined efforts of agencies, stakeholders, and communities (NIST 2021).

In practice, the decision to design a building to meet recovery-based objectives will depend on costeffectiveness, which has been identified by stakeholders as the most important attribute in assessing and implementing functional recovery options (Abrahams et al. 2021). FEMA's Benefit-Cost Analysis Guide (FEMA 2021) and ASTM-E1074 Standard Practice (ASTM International 2020a-f) all provide guidelines for benefit and cost evaluation on conventional building improvements. However, there is no standard approach for the economic evaluation of functional recovery design, which is a new concept itself. Therefore, a new framework is needed to assist decision makers to evaluate the cost-effectiveness of adopting recovery-based design criteria.

The goal of this report is to provide actionable information and best-practice guidance tailored to decision makers at various scales (individuals, community, and government). The present framework includes the methods to assess the benefits and costs of building design beyond code requirements, and the approaches to quantify the value of enhanced recovery of building functions after a natural hazard event. In particular, ten loss categories and twenty-seven benefit categories are presented to support decision making of different stakeholder groups. Three recovery states (i.e., reoccupancy, functional recovery, and full recovery) that prioritize building functions and repair sequences are incorporated into the framework for economic evaluation on business interruption loss.

Two main challenges associated with benefit-cost analysis of recovery-based design are addressed in this report: (1) Functional recovery may affect various stakeholders with conflicting goals (e.g., those who can pay for functional recovery are not necessarily the same as those who benefit from it). We therefore identify impacts at three distinct tiers: individuals (building owner, developer, occupants); community (non-occupants who benefit from a building's services); and government (local, state, and federal). At each tier, the framework identifies direct and indirect benefits and costs, as well as positive and negative externalities, potential co-benefits, and sources of uncertainty. (2) Additionally, functional recovery is a new and evolving concept, and design criteria do not yet exist (ICC 2019). Thus, what it means, and what it costs, for a building to be designed to provide functional recovery is not well-defined. To circumvent this gap, we illustrate the estimation of construction costs for a range of building designs that achieve comparable post-earthquake performance.

The key takeaways of this report are as follows:

• Compared to traditional code-conforming design, recovery-based design provides additional benefits to building owners, occupants, and the community by reducing repair costs, displacement costs, business interruption losses, and rental losses after a seismic event.

- Improving seismic performance of nonstructural components can significantly enhance building overall performance and can produce greater net benefits compared to structural improvements only.
- It is important to understand how benefits and costs are disturbed among various stakeholders because the amount paid are often not proportional to the amount of benefit enjoyed.
- While direct benefits from mitigation actions are well studied (e.g., avoided building damage and human injuries), indirect benefits are often ignored in benefit-cost analysis. However, these indirect benefits can be highly relevant to community resilience, social equity, and environmental sustainability. In addition, the data required to evaluate indirect benefits is scarce even though the methods are readily available.

While this report is focused on seismic hazards, the concept of functional recovery is applicable to other natural and human-made hazards, and many of the contributions of the proposed BCA framework can be extended to additional hazards.

This report does not provide technical guidance on design standards, specifically on how to define and design functional recovery performance targets. Rather, the goal of this report is to provide a risk-based economic framework for selecting among candidate design options, as well as how building owners, occupants, and communities can evaluate the decision of whether to adopt a new functional recovery design standard. On the other hand, the framework can be used in conjunction with development of design standards if there is a desire to ensure economic feasibility of a candidate design standard. The goal of the report is to build on existing economic analysis tools by providing a catalog of inputs (and methods for estimating them) for such tools. Our contribution is providing the roadmap for the inputs specific to an economic analysis of functional recovery design, as well as available tools and measurement needs.

The report is organized as follows:

- Section 2 provides background information on functional recovery
- Section 3 reviews options for the economic evaluation of building design standards
- Section 4 reviews the literature on conducting benefit-cost analysis for building design standards
- Section 5 presents the framework for Functional Recovery Benefit-Cost Analysis (FR-BCA), including a brief case study example to illustrate the FR-BCA framework
- Section 6 reviews remaining gaps and highlights directions for future research

2. Background on Functional Recovery

2.1. Development of seismic building codes and standards

The extent of seismic impacts within the United States have been well-demonstrated through the New Madrid (1811), Charleston (1886), San Francisco (1905), Long Beach (1933), Alaska (1946 and 1964), San Fernando (1971), Loma Prieta (1989), and Northridge (1994) earthquakes. Each produced lessons on risks from buildings and infrastructure, fires, liquefaction, and tsunami. Even with some of the earliest of these earthquakes, communities have often modified building codes, standards, and practices after an event to help reduce destruction for the purposes of preserving lives.¹ Historically, building codes and standards have been agreed upon and enforced because they are viewed as minimum safety standards (Zhang et al. 2021).

The International Building Code (IBC), first published in 2000, is a model code widely adopted by states and local jurisdictions for building design, construction, maintenance, repair, and demolition (ICC 2021). The code provides minimum requirements to safeguard public health, safety, and welfare, which ensures buildings meet the life-safety performance objective.² The ASCE 7-16 standards for *Minimum Design Loads and Associated Criteria for Buildings and Other Structures* form the basis of the structural provisions of the IBC (ASCE 2017a). Other standards referenced by IBC include ACI 318 *Building Code Requirements for Structural Concrete* (ACI 2021), TMS 402/602 *Building Code Requirements and Specification for Structural Masonry* (TMS 2021), AISC 360 *Specification for Structural Steel Buildings*, and the *National Design Specification (NDS) for Wood Construction* (AISC 2021).

In ASCE 7-16, critical buildings such as hospitals and schools are designed to a higher level of performance compared to ordinary buildings. The ASCE 7-16 standards classify buildings into four categories based on the risk posed by the building in the event of failure (See Appendix A). The four risk categories are reflected by the importance factor, a modifier for design lateral force, in seismic design. Commercial and residential buildings (Risk Category II) use an importance factor of 1, while schools (Risk Category III) and hospitals (Risk Category IV) use a higher importance factor of 1.25 and 1.5 respectively, enabling them to sustain 25 % and 50 % larger seismic forces. Moreover, schools and hospitals are subject to tighter drift limits, which helps to minimize damage to some structural and non-structural components (Ghosh 2019).

While current building codes are intended to prevent building collapse and ensure people can evacuate safely, buildings that meet these minimum safety requirements may still sustain severe damage and be unrepairable or unusable after an earthquake. Interruptions to the operation of critical facilities can cause widespread social and economic loss as demonstrated by the earthquakes at the end of the 20th century (e.g., the 1994 Northridge and 1995 Kobe Earthquakes). As a result, some states like California adopted legislation to address the need for adequate protection of hospitals and other critical facilities. For instance, California Senate Bill 1953, signed into law on September 21, 1994, requires all hospitals to be capable of operating following a large earthquake by 2030 (Meade and Kulick 2006).

¹ An example is the increased earthquake design for California schools instituted after the 1933 Long Beach earthquake (Turner 2004). ² The IBC applies to nearly all types of new buildings. The International Residential Code (IRC) applies to new one- and two-family dwellings and townhouses of not more than three stories in height. The International Existing Building Code (IEBC) applies to the alteration, repair, addition, or change in occupancy of existing structures. See <u>https://www.fema.gov/emergency-managers/risk-management/earthquake/seismic-building-codes</u> for more details.

2.2. Performance-Based Earthquake Engineering

Alternative to traditional code-based design, performance-based design (PBD) offers a building design philosophy in which the performance of a particular design is analytically quantified and compared with target performance metrics, set by building owners or other stakeholders, to determine the acceptability of the design. Target performance metrics can be tailored to the needs of a specific project or be adopted from a performance-based standard; however, minimum requirements specified by building codes should be fulfilled. Common performance metrics include repair costs, downtime, casualties, as well as a discrete performance state classification (e.g., Immediate occupancy, Collapse Prevention). The most widely adopted PBD standards and methods in the U.S. include ASCE 41 (ASCE 2017b), the PEER Tall Building Initiative (PEER 2017), and FEMA P-58 (FEMA 2018). While PBD offers advantages in terms of the explicit quantification and communication of expected building performance, it is often more engineering intensive and computationally expensive compared to code-based design. In the United States, most buildings are designed based on locally adopted building codes, while PBD methods are used in seismic retrofits and design of tall buildings in high seismic hazard zones.

2.3. What is functional recovery?

As modern building performance has improved, recent earthquakes have pointed out additional needs for quicker recovery for buildings and key lifeline infrastructure systems. Enhanced performance is needed to lessen economic impacts (Kroll et al. 1991), prevent catastrophic disruption to transportation systems (EERI 2019, Yashinsky 1998), ensure water and fuel availability (Taylor 2014), and reduce vulnerability of critical buildings (Jaret 2019). The 2010-2011 Canterbury earthquake sequence in New Zealand, in particular, demonstrated the potential for extensive and long-term effects across all facets of an urban society (Potter et al. 2015). In addition, advances in science, engineering, and cross-disciplinary collaboration have improved the robustness and applicability of regional earthquake damage simulations to illustrate potential community impacts to residents, planners, and politicians (USGS 2021).

While improving aspects of recovery from earthquakes is not new (Beck et al. 1997, Comerio 2014, REDi 2013) attempts to support those goals via the alteration of building codes, standards, and practices has increased in the past few years. This shift towards improving post-earthquake outcomes changes the emphasis from minimal life safety objectives to a higher state of performance linked to specific recovery needs. In California, recent bills attempted to promote higher performance levels and to identify particularly vulnerable buildings (EERI 2021). The National Institute of Standards and Technology (NIST), as well as the Federal Emergency Management Agency (FEMA), two federal agency members of the National Earthquake Hazards Reduction Program (NEHRP), recently received charges from Congress to lead efforts in the area of improved recovery.

NIST SP1224 (NIST 2018) details findings for research needs and implementation actions to produce an immediate occupancy performance objective across natural hazards. More recently, NIST and FEMA published FEMA P-2090/NIST SP1254 (NIST 2021), a report containing options for improving the built environment and critical infrastructure to reflect performance goals stated in terms of post-earthquake re-occupancy and functional recovery time.

FEMA P-2090/NIST SP1254 recommends developing codes and standards to consider functional recovery design objectives for new buildings, existing buildings, and lifeline infrastructure, in addition to pre-disaster planning, education and outreach, and access to financial resources.

As NEHRP's advisory committee points out: "Designing new buildings and retrofitting existing buildings to a functional recovery design objective will better align with public expectations regarding seismic performance of the infrastructure, enable our communities to recover more quickly following an earthquake, and ultimately achieve the resilience desired." (ACEHR 2019). Figure 1 illustrates the theoretical range of building performance attainable for recovery-based design objectives.



Figure 1. Theoretical range of building performance and relative placement of safety-based and recoverybased goals. Source: FEMA P-2090/NIST SP1254.

In the figure, recovery-based goals are placed relative to safety-based goals. Of the recovery-based goals, reoccupancy represents a building performance state that is safe and habitable and can be used as basic shelter, whereas Full Functionality occurs when the building is at, or restored to, its pre-earthquake condition. Functional recovery is a performance state higher than reoccupancy, but lower than full functionality, targeting reasonable recovery times for basic re-occupancy and service provisioning in order to reduce displacement, downtime, and significant long-term social and economic impacts.

NIST and FEMA's publications are the result of state-of-the-art thinking by many individuals and organizations across various fields of expertise, including: structural engineering, civil engineering, architecture, urban planning, emergency management, code development, disaster science, economics, and public policy. While the general goals for functional recovery are quite clear, the mechanisms to put those goals into practice within buildings and lifelines infrastructure needs further work (ICC 2020, NIST 2021). Translating ideas for improvements to resilience and performance into the technical and practical approaches necessary will require long-term commitment, collaboration and consensus-building.

The framework presented in this report for assessing costs and benefits related to improved performance represents one key component for further development. While critical lifelines are fundamental for the reoccupancy and recovery of a building, economic assessment is more challenging due to the spatial

distribution of lifeline networks. Thus, in this report we focus on functional recovery of buildings and adopt the FEMA P-2090/NIST SP1254 definition of functional recovery specific to earthquakes:

Definition: Functional recovery is a post-earthquake performance state in which a building is maintained, or restored, to safely and adequately support the basic intended functions associated with the pre-earthquake use or occupancy of the building.

The economic evaluation of recovery-based design standards for critical infrastructure is left for future work.

3. Options for the Economic Evaluation of Recovery-Based Design

In this section, we review methodological options for the economic evaluation of improved seismic design standards. The options vary in their scope and data requirements. It is worth noting that the options presented are not mutually exclusive, and in some cases may be complementary. Moreover, the methods covered in this section are not exhaustive, but meant to be representative of the range of options available.

Note that in principle, while the focus of this report is on providing the tools to evaluate alternative candidates for recovery-based design standards ("adoption"), that decision will be driven by evaluating the economic benefits and costs associated with actually designing or retrofitting a building to a new standard ("implementation"). Thus, the economic evaluation is from the perspective of those who implement the design (typically developers and building owners), as well as those who are affected by the design (occupants, neighbors, the community), rather than those that set design codes and standards (state and local governments, independent entities, federal agencies).

3.1. Benefit-cost analysis

The idea behind benefit-cost analysis (BCA) is simple: do the outcomes of an action outweigh the costs of investing in that action? The outcomes of interest include all of the possible direct, indirect, and intangible benefits of an action (Boardman et al. 2017). The key to BCA is that benefits and costs are measured in the same units so that they are comparable. In particular, benefits that may be nonmonetary in nature are typically converted to monetary values (Zerbe and Bellas 2006).

The goal of BCA is to capture a comprehensive accounting of potential benefits and costs over time, regardless of when they occur (Boardman et al. 2017). Depending on the action, benefits may continue accruing beyond the length of the investment (e.g., public health interventions). For actions that affect the built infrastructure, including new construction, demolition, and renovation, economic evaluation is restricted to the lifetime of the project (e.g., a building's useful life). In any case, a discount rate, δ , is used to normalize the stream of future benefits to present value terms.

The most familiar evaluation criterion consists of a straightforward ratio of all the possible benefits of the action to the costs of undertaking the action, the benefit-cost ratio (BCR):

$$BCR = \frac{benefits}{costs} \tag{1}$$

If the BCR > 1, the action may be recommended on the basis that the benefits outweigh the costs. When evaluating or ranking a *set* of alternative actions, a decision may be justified on the basis of choosing the action that maximizes the BCR. However, when choosing between alternative design standards for a given building, it is recommended practice to consider the incremental benefits gained relative to the additional cost of each design, otherwise relative BCRs may not reflect the relative benefits of the alternatives (ASTM E964). For instance, there may be different operating and maintenance costs ("operational costs") associated with each design, which are distinct from the costs of implementing the design ("investment costs") and complicate the comparison of the numerator to the denominator:

$$BCR = \frac{(benefits - operational costs)}{investment costs}$$
(2)

A major caveat to the BCR > 1 criterion is that there may be other, perhaps unquantifiable or non-economic factors that can affect a decision to adopt an action, such as fairness, public opinion, and political factors (Arrow et al. 1996; Tulchinksky and Varavikova 2014). It is worth noting that a benefit-cost analysis is only one component of a risk mitigation strategy, which includes performing a risk assessment, specifying combinations of engineering, management, and financial risk mitigation strategies, and performing an economic evaluation for the portfolio of strategies (ASTM E2506).

Note that because the BCR is a ratio of dollars to dollars, the metric has no units. In practice, the BCR is often interpreted as saying: "Every dollar we spend on X will save Y dollars," where X is the action and Y is the BCR (e.g., Miller and Hendrie 2008; NIBS 2019).

On the other hand, once benefits and costs are quantified, a decision maker can consider alternative metrics than the BCR.³ For example, the net present value (NPV) of an investment is simply the (discounted) benefits net of (investment) costs:

$$NPV = benefits - costs \tag{3}$$

In this case, the criterion for investment is $NPV > 0.^4$ Note that NPV is measured in dollars, in contrast to the BCR. Thus, the metric is naturally suited to comparing or ranking a set of alternatives and is not subject to ambiguities due to different operational costs.

Typically, BCA is prospective (conducted ex ante), in order to evaluate the expected value of a potential intervention; e.g., environmental protection (Pearce et al. 2006); management of health care (Tulchinsky and Varavikova 2014); and the evaluation of education policies (Fletcher 2010). However, BCA can also be retrospective (conducted ex post) to analyze the impact of an actual policy action, such as the impact of enhanced building codes in Moore, OK in response to destructive tornadoes (Simmons et al. 2015). There is also potential to conduct BCA both ex ante and ex post when evaluating the value of improving an existing policy, such as expanding investments in COVID-19 vaccine capacity (Castillo et al. 2021).

The simplicity of BCA belies the potential complexity of estimating the benefits. The challenge lies in not only enumerating the set of potential benefits, but also in monetizing those benefits; for example, monetizing the value of improving public health through upgrades to water and wastewater infrastructure systems (Zerbe and Bellas 2006). On the other hand, where possible, the ability to monetize benefits and compare all benefits and costs measured in the same units is also a strength of BCA. For the evaluation of design standards whose objectives are to improve building performance relative to some hazard, the direct benefits are the avoided losses from the action (e.g., reduced damage from a hazard due to higher design standards). Estimating avoided losses thus requires evaluating building performance under stress. For hazards such as earthquakes, this is typically estimated using software tools such as Hazus (e.g., NIBS 2019) and OpenSees (e.g., Goulet et al. 2007). Once avoided losses are estimated, they can be entered into benefit-cost analysis software, such as FEMA's BCA tool (FEMA 2021) or NIST EDGe\$ (Helgeson et al.

 $^{^{3}}$ It should be noted that the calculated BCR may result in ambiguities (e.g., if BCR < 0, is it because the numerator is negative or the

denominator?). The validity of the BCR may be questioned, in particular as regards to the manipulability of the numerator, so it is important to be as clear as possible about all of the underlying assumptions. As with any other metric, it is meant as a guide rather than an absolute rule.

⁴ This criterion is also known as the net benefits (NB) criterion (ASTM E1074).

2020).⁵ Section 4 discusses methods and software tools used in BCA studies of design standards. The application of BCA to recovery-based design is described in more detail in Sec. 5.

Alternative criteria include the Savings to Investment Ratio (SIR),⁶ the Internal Rate of Return (IRR),⁷ and payback (PB) or break-even analysis.⁸ In addition, there are related metrics such as life-cycle costs (LCC, discussed in Sec. 3.3) and the Return on Investment (roughly, ROI = NPV/costs) that may be more appropriate depending on the action. More details on discounting and alternative metrics for actions specific to buildings and building investments are provided in ASTM E1185. Table 3.1 provides a summary of benefit-cost related metrics and their use for building investment decisions. As the table illustrates, BCA using BCR or NPV can be applied across a broad range of building investment decisions, which include:

- Accept/Reject: a single choice (e.g., design to higher standard or not)
- Design: choosing one out of several competing design options (e.g., choosing between several candidates for a recovery-based design standard)
- Size: the choice of scale or level of investment (e.g., how much bracing)
- Priority or ranking: choice of one or more options from a group (e.g., what parts of building to renovate)

Table 3.1. Alternative metrics for building investment decisions and their use for different types of decisions. Adapted from ASTM E1185.

Decision	BCR (or SIR)	NPV (or NS)	IRR (or AIRR)	PB	LCC
Accept/Reject	Х	Х	Х	X**	Х
Design	X*	Х	X*		Х
Size	X*	Х	X*		Х
Priority or ranking	Х	Х	Х		

Note: X = Acceptable; $X^* = Acceptable$ if using incremental benefits and costs; $X^{**} = Acceptable$ with caveats.

3.1.1. Extensions of "traditional" BCA methods

Traditional BCA attempts a holistic approach to evaluating outcomes for all relevant stakeholders (e.g., building owners and tenants). For these reasons, BCA is sometimes referred to as *social* benefit-cost analysis (Boardman et al. 2017). However, in certain applications it might make more sense to focus on BCA from the perspective of specific stakeholders. For example, mitigation decisions such as seismic

⁵ It is worth noting that the methodology used in the EDGe\$ software is based on a guidance document (Gilbert et al. 2015) that is the basis for a standard for the economic evaluation of resilience strategies (ASTM E3130).

⁶ The SIR differs from the BCR in that it prioritizes cost savings of an action, where cost savings are defined as benefits net of operation and maintenance costs (not investment costs). For building projects, it is assumed that total operation and maintenance costs over the life of the project are less than zero (i.e., the project saves costs) and moreover that total cost reductions exceed total benefits (ASTM E964). ⁷ The IRR is defined as the discount rate such that NPV = 0 (ASTM E1057; ASCE 2021).

⁸ The payback period is defined as the length of time, typically in years, until the NPV is equal to the initial investment (ASTM E1121).

retrofits are typically made by the building owner. Thus, Cutfield and Ma (2015) present a building-owner focused BCA that shifts the perspective to the owner to assess key drivers in the decision to retrofit, demolish, or do nothing. The analysis places emphasis on all of the owner's potential cash flows and includes the option to sell the building, while ignoring social benefits that would not persuade the building owner to take action.

A different perspective builds on the observation that investing in an action has diminishing returns: an additional dollar invested does not necessarily result in a proportionate level of benefits. A core concept in economics is that net benefits are maximized when the *marginal benefit* is equal to the *marginal cost*, MB = MC (Arrow et al. 1996). Marginal BCA can thus be used to choose from a set of options by optimizing the marginal benefit relative to the marginal cost. For example, Li et al. (2009) adopt an expected utility framework to conduct a marginal BCA over four seismic retrofit options in Turkey. While the application in Li et al. (2009) ranks alternative options, marginal BCA is best suited to problems where a decision maker such as a regulator must optimize the *level* of risk reduction across a set of options (size decisions). Thus, a key challenge with this approach is in measuring the level of risk reduction relative to monetary benefits and costs.

Finally, a very different extension is to expand the scope of the BCR from a single number to a range that captures some of the uncertainty inherent in estimating benefits. Probabilistic BCA models the distribution of the BCR in order to derive exceedance curves for the probability that BCR > 1. Cardona et al. (2008) model the net present value of losses as a random variable and thus derive analytical formulas for the probability distribution of the BCR. In addition, Mora et al. (2012) demonstrate a Monte Carlo computation of the distribution based on the loss exceedance curve produced by standard probabilistic risk analysis that closely approximates the analytical form. Similarly, Ghesquiere et al. (2006) model annual losses as a random variable and derive BCR exceedance curves using outputs from an earthquake risk assessment of buildings in Bogota, Colombia.

3.2. Cost-effectiveness analysis

An alternative approach to benefit-cost analysis is cost-effectiveness analysis (CEA). In contrast to BCA, the objective of CEA is to choose the option that minimizes the cost of an action relative to a *single outcome* of interest; e.g., reduced casualties from seismic codes (Peterson and Small 2012) or curriculum changes to improve student achievement (Levin et al. 2003). It is therefore used to determine the least expensive way of achieving a particular, well-defined goal (Tulchinsky and Varavikova 2014). The advantage of CEA is that the analysis does not require quantifying the full range of benefits. Indeed, a decision maker may simply optimize a single measure without consideration of benefits at all. For example, Nuti and Vanzi (2003) develop a parametric criterion to evaluate the decision to retrofit a building, based on the ratio of retrofit cost to the repair cost and the change in mean failure rate due to retrofit. The criterion is derived by minimizing the annual equivalent cost, an alternative to net present value.⁹

A disadvantage of CEA is that it implicitly assumes some action to be undertaken to achieve a desired outcome. On the other hand, BCA typically considers the value of an action relative to the status quo (or "do nothing") option (Zerbe and Bellas 2006). One potential advantage of conducting a BCA is that by enumerating the range of potential costs and benefits from an action, one can also conduct a CEA by

⁹ Annual Equivalent Cost, more commonly known as Equivalent Annual Cost (EAC), is a method that converts cash flows from an asset to uniform, annual amounts (Jones and Smith 1982). In contrast, Net Present Value (NPV) discounts cash flows to present value terms.

focusing on a single outcome (e.g., reduced downtime) and comparing cost across a set of options for achieving that outcome. However, the outcome must be equivalent across options. In the case of recovery-based design, for instance, a CEA would compare costs of alternative design options that each achieve re-occupancy within three weeks (with 90 % probability under a design-level earthquake). In addition, tools like NIST's EDGe\$ can conduct both BCA and CEA at the same time (Helgeson et al. 2020).¹⁰

3.3. Life-cycle cost analysis

Life-cycle cost (LCC) analysis, like CEA, focuses on a project's cost-effectiveness. However, the key difference is that LCCs cover all relevant (discounted) project costs, including operation and maintenance costs as well as any other costs incurred for the assessment period (e.g., acquisition, ownership, operational costs, and disposal) (ASCE 2021). It is analogous to BCA, but the focus is on costs and the only benefits considered are those that are realized as lower costs (or savings).

As shown in Table 3.1, life-cycle cost analysis is applicable to accept/reject, design, and size decisions that are driven by cost (ASTM E1185). However, it is important to note that comparing alternatives requires the alternatives to satisfy the same functional requirements across the same study period (ASTM E917). The LCC method is especially suited to potential actions for which higher initial costs may be justified by lower future costs (ASCE 2021). For instance, LCC is routinely used for the evaluation of investments in building energy efficiency, as the potential cost savings from such investments are key to investment decisions (Kneifel 2010).

Life-cycle costs may be useful for quantifying the expected damage from earthquake events across the life of the building (Ramirez et al. 2012). For earthquake risk reduction, LCC analysis has been applied to the evaluation of the optimal design of buildings (Wen and Kang 2001; Liu et al. 2003); the optimal level of strengthening, conditional on undertaking a retrofit (Kappos and Dimitrakopoulos 2008); and the management of highway bridge systems (Frangopol et al. 2001). In all cases, the primary consideration is a design or size decision in which the stream of costs over the life of a structure is important. In addition, LCC analysis has potential applications for the evaluation of programs that pair energy efficiency with earthquake risk reduction as documented in Zhang et al. (2022). For recovery-based design, LCC may be useful for comparing alternative design options that all deliver the same performance (in terms of reoccupancy and recovery of function targets) and initial upfront costs may not tell the whole story (for example, if one has higher initial costs but lower maintenance costs).

3.4. Willingness-to-pay

In economics, willingness-to-pay (WTP) is a measure of how much an individual would pay for a good or service—typically one that is not traded in a market and therefore does not have a market price, such as clean air or a potential new product (Breidert et al. 2006).¹¹ For earthquake risk reduction, for example, it may be used to directly estimate the value homeowners place on seismic strengthening (Manganelli et al. 2018). Willingness-to-pay is sometimes used in policy or legal arguments as a proxy measure for the welfare an individual derives from a good or service (Bar-Gill 2020). However, the relationship between

¹⁰ An alternative that does consider the status quo is the incremental cost-effectiveness ratio (ICER), which is the ratio of the cost difference of an action relative to the status quo to the differential effectiveness between the action and the status quo, where effectiveness is with respect to a single measurable outcome (Bilger 2017). The ICER is commonly used in the medical literature to evaluate health interventions.

¹¹ An alternative measure, willingness-to-accept (WTA), represents how much an individual would have to be compensated for a good or service.

WTP and welfare is tenuous; for instance, people are often willing to pay for goods or services that do not improve their welfare (Sunstein 2007).

Most often, WTP is used to monetize benefits from an action that are difficult to price. For example, the value of a statistical life (VSL), which is used to monetize the value of reduced mortality risk from a policy, is based on an aggregate measure of individual estimates of WTP (Sunstein 2013). In this way, WTP may be used as a complement to BCA and has become a standard method for valuing ecosystem services and other environmental amenities (Stevens et al. 2000). For earthquake risk reduction, WTP may be used to approximate how much an individual values enhanced building codes by focusing explicitly on the WTP for the benefits derived from improved performance, such as an increase in the building's useful life or a reduction in environmental impacts (Di Bari et al. 2020; Belleri and Marini 2016).

There are two main approaches to measuring WTP (Breidart et al. 2006):

- Stated preference methods (e.g., contingent valuation, conjoint analysis) are based on surveys or interviews that require respondents to evaluate hypothetical scenarios, or "choice experiments" (Kanya et al. 2019). Willingness-to-pay is estimated by either explicitly asking people for their WTP for a good or service or implicitly having them value related goods or services. However, stated preference estimates of WTP are often criticized as suffering from hypothetical bias (differing from actual WTP), strategic bias (respondents want to please the researcher), and for high variation in WTP estimates across methods (Stevens et al. 2000). Thus, stated preference methods may be best used as an approximate lower bound for WTP.¹² Finally, it should be noted that, as with survey methods in general, stated preference methods for estimating WTP may be time consuming and costly, reducing their practical value.
- Revealed preference methods (e.g., hedonic analysis, travel cost method) infer WTP from pricing decisions, either from actual markets for goods or experimental designs to simulate market trade. Thus, to the extent that revealed preferences align more closely with a consumer's actual willingness to pay for a good or service, the method requires data on market transactions. This requirement constrains the scope for the use of revealed preference methods. It should be noted, however, that a strength of revealed preference methods is that the good or service need not be itself tradeable in a market; e.g., hedonic methods are routinely used to estimate willingness-to-pay for urban and environmental amenities such as school quality (Black 1999), clean air (Chay and Greenstone 2005), and the remediation of hazardous waste sites (Greenstone and Gallagher 2008). Of course, the major caveat to revealed preference methods is the requirement of (backwardslooking) data or the careful simulation of a market setting.

Most methods focus on measuring *use value*, which is the value derived from actual use of or interaction with the good or service. A few methods, such as contingent valuation, can also measure *non-use value*, which is not tied to direct use of the good or service, but is rather value derived from the existence of a good or service; e.g., an individual may place a positive value on an aquatic ecosystem even if the individual derives no direct benefit from it (Kanya et al. 2019). While non-use value may be relevant for recovery-based design, it is unlikely that it is a significant enough fraction of the benefits.

¹² WTP is also criticized for being sensitive to wealth effects; that is, individuals face different budget constraints and value goods accordingly. In some cases, WTA may be a more reliable measure to the extent that it is less sensitive than WTP to wealth effects (Bar-Gill 2020).

3.5. Economic impact analysis

A very different approach to economic evaluation is economic impact analysis (EIA). As the name suggests, the main purpose of EIA is to determine how an action affects economic activity (e.g., employment, income). A distinguishing feature of EIA is the focus on a well-defined region of economic activity (Weisbrod et al. 2016). In contrast, BCA focuses on determining whether an action makes "society" better off, typically without a well-defined geographic boundary.¹³ In addition to the spatial dimension, BCA and EIA differ on the following dimensions:

- The temporal dimension: BCA focuses on the present value of impacts within a well-defined study period, while EIA focuses on undiscounted future impacts within a less well-defined medium and long-term.
- The impact dimension: EIA focuses on impacts on money flows (such as expenditures); in principle, BCA considers all benefits and costs to society, but as the discussion in Sec. 3.1 illustrates, in practice this is often reduced to those that can be monetized.¹⁴
- The output: while BCA summarizes each of the impacts into a single metric (BCR or NPV), the impacts in an EIA (e.g., employment, income) remain disaggregated as the objective of EIA is to determine how impacts are distributed in the economy.

Thus, BCA and EIA may be seen as complementary approaches with different perspectives. For instance, EIA could be used to measure impacts on productivity or employment, which could then be used as inputs into a BCA.

Economic impact analysis has been used to explore changes in transportation policy (Weisbrod et al. 2016), regional impacts of large events such as sporting events (Dwyer et al. 2006), and to quantify the direct and indirect impacts of natural disasters (Okuyama 2007). The standard approach to economic impact analysis relies on input-output (I-O) models, which rely on accounting tables that trace inter-industry purchases and sales in a region (Weisbrod and Weisbrod 1997). An I-O models economic multipliers that determine how a dollar spent on a good or service leads to direct and indirect impacts on employment, income, and output.

One criticism of this approach is that it focused solely on positive impacts and ignores negative impacts (Taks et al. 2011). Simulation models, such as computable general equilibrium (CGE) or agent-based models, build on I-O models by considering additional aspects of the economy such as household purchases from industry, shifts in population, and productivity (Weisbrod and Weisbrod 1997; Hallegatte 2008; Inoue and Todo 2019). Such models therefore provide a more comprehensive picture of economic impacts. Of course, the added level of sophistication comes with greater modeling and data costs, making model assessment by non-experts is difficult.

At the margin, designing or retrofitting a single building for recovery likely has a negligible impact on the local economy and thus BCA is the more appropriate method for economic evaluation. On the other hand,

¹³ It should be noted that while the intent is for BCA to take a holistic perspective, in practice it is challenging to fully account for the complexities of society, including who bears the benefits and costs and associated equity concerns.

¹⁴ Moreover, BCA implicitly assumes (a) independent valuations for each impact that reflect trade-offs among the different types of benefits and costs, and are additive for calculating total impacts over time (Weisbrod et al. 2016). In general, EIA does not depend on these assumptions.

economic impact analysis may be useful for quantifying the direct and indirect benefits¹⁵ of communitylevel adoption of enhanced building design.¹⁶ Moreover, such analyses might provide reasonable bounds on the indirect impacts of recovery-based design, such as reduced supply chain disruptions, that could be used as inputs into BCA.

Due to the nature of EIA models, it may be difficult to distinguish spending flows as benefits or costs. In cases where this distinction is important, BCA may be the preferred approach. Finally, it is important to note that by design EIA is not intended to provide outputs or other information that can inform a decision support tool, while BCA does so.

3.6. Summary

For the purposes of recovery-based design, the preceding discussion suggests BCA as the most appropriate method because it is comprehensive and is particularly well-suited to decision support for investments in single buildings. However, other methods such as CEA and LCC may be appropriate if the primary concern is the cost side.

¹⁵ In principle, EIA can also be used to estimate co-benefits, which are benefits that occur even if there is no hazard (Fung et al. 2021a).

¹⁶ For instance, suppose enhanced building designs are more attractive; they may increase property values, which attracts wealthier

businesses/tenants, which increases the tax base, which increases public services, which increases quality of life, which attracts even more people.

4. Benefit-Cost Analysis for Building Design: A Review

Economic evaluation of enhanced building design, such as benefit-cost analysis, does not typically consider above code design, but rather is conducted contemporaneously with newly proposed or implemented code changes. Much of the focus is therefore on designing new buildings or bringing existing buildings up to code. This section presents a review of the literature, with a particular focus on methodologies and tools used. The review is by no means comprehensive, but is meant to be illustrative of the literature. For an extensive review of the state-of-the-art, see Zhang et al. (2022). The main takeaway is that functional recovery represents a new design paradigm and these earlier studies do not provide an adequate precedent for the economic evaluation of recovery-based design.

4.1. Benefits and costs of adopting higher standards for new construction

For new buildings, studies are typically conducted to analyze the economic impact of code changes; in particular, the impact of complying with modern building codes. Not all studies conduct a full benefit-cost analysis. Some focus on the benefits (reduction avoided losses) associated with compliance, while others focus on the additional construction cost associated with compliance. A few exceptions (Goulet et al. 2007; NIBS 2019) consider exceeding code requirement, as discussed at the end of this subsection.

Ryu et al. (2010) compare annual expected losses for commercial buildings constructed to a life-safety performance objective using 2003 IBC, 2006 IBC, and 1999 Standard Building Code (SBC) in Memphis, TN. Using Hazus, they find that losses can be reduced by approximately 1 % when either 2003 or 2006 IBC is implemented, relative to 1999 SBC. FEMA (2020) conducts a study of the value of meeting I-codes for new construction for 2000-2018, relative to 1994 UBC. Using damage functions from Hazus with parcel and building footprint data for six western states,¹⁷ as well as input from experts in building performance and building code history, FEMA (2020) estimates roughly \$60 million in avoided losses associated with complying with I-codes.

The National Association of Home Builders (NAHB) assesses the costs for single and multi-family homes conforming to 2018 IBC, using a combination of RSMeans Cost data, Census data, Bureau of Labor Statistics data, and data from distributors' or retailers' websites (NAHB 2018). For multi-family homes, the construction cost is expected to increase by around \$16 000 for a two-story apartment and by around \$44 000 for a three-story apartment when subject to the code change of 2018 IBC, relative to 2015 IBC.

NEHRP Consultants Joint Venture (2013) employs the PEER Performance-Based Earthquake Engineering (PBEE) design framework, implemented in FEMA's PACT software (FEMA 2012b), to analyze the impacts of compliance with 2012 IBC, relative to 1999 SBC. Cost estimates are developed by a consulting firm. Results indicate that the structural cost of a two-story reinforced concrete office building (total floor area of 100 000 square feet) may increase by 4.6 % when designing for 2012 IBC, due to increased requirements for braced frames, collectors, and foundations. The total building cost may grow by 0.7 % when adding required bracing and anchorage of nonstructural components and systems to the building. These measures together can reduce annualized losses by 40 %.

¹⁷ FEMA (2020) focuses on the six states with the highest seismicity (Alaska, California, Hawaii, Oregon, Utah, and Washington), which together account 78.5 % of the national Average Annualized Losses per FEMA (2017).

The National Institute of Building Sciences (NIBS) analyzes the benefits and costs of implementing 2018 IBC seismic design requirements for the minimum performance objective of life safety (NIBS 2019). Using Hazus, they estimate annualized losses due to earthquakes for one percent of the building inventory across the 48 contiguous United States. Compared to 1990's seismic codes, NIBS (2019) finds that 2018 IBC can help prevent property losses of \$1500 per building, reduce deaths, injuries, and trauma-related losses by \$800 per building, and decrease business interruption losses by \$2000 per building. The total benefit is expected to be \$4.3 billion per year, three times greater than the total cost (i.e., BCR of 3). The construction cost is assumed to increase by 1 % for a 50 % increase in strength and stiffness (based on Porter 2016). Since the strength requirement of the 1990's code is approximately 67 % of that of 2018 I-code, the increased compliance cost for 2018 I-code is 0.7 % (NIBS 2019).¹⁸

NIBS (2019) also creates a benefit-transfer matrix, shown in Table 4.1, to allocate the estimated benefit to five closely involved stakeholder groups: developers, title holders, lenders, tenants, and communities. Among the five groups, tenants benefit most from the code with a net benefit up to \$2 billion, followed by title holders, communities, and lenders. The assumptions underlying the benefit-transfer matrix represented in Table 2 are derived based on a process for eliciting expert judgment (NIBS 2019).

Stakeholder group	Construction cost	Property loss	Direct business interruption	Indirect business interruption	Insurance	Death and injury
Developer		2 %			4 %	
Title holder	50 %	58 %			86 %	
Lender		7 %			10 %	
Tenant	50 %	33 %	100 %			99 %
Community				100 %		1 %

Table 4.1. Benefit-transfer matrix (NIBS council 2019).

On the other hand, the literature on evaluating benefits and costs of above-code seismic design is more limited. Goulet et al. (2007) evaluate the benefits of designing reinforced concrete buildings to exceed 2003 IBC seismic design requirements. The results show that above-code design can reduce expected annual losses from fatalities by up to 66 % compared to code-minimum design. Particularly, using uniform beams and columns throughout the building can lower expected annual losses by 22 % compared to the original design.

¹⁸ The construction cost for buildings complying with the 1990's code is estimated using the RSMeans CostWorks 2018 data.

NIBS (2019) assesses the cost-effectiveness of designing buildings to exceed 2015 IBC strength and stiffness requirements. The study assumes that 1 % of the existing building inventory is replaced by new buildings that are designed for a higher importance factor of 1, 1.25, 1.5, 2, or 3, varying by county.¹⁹ The tabulated vulnerability functions are used to estimate annualized losses of structures and nonstructural contents due to earthquakes. For construction cost, the same assumptions are made as for compliance with 2018 IBC; that is, the increased construction cost is 1% due to a 50% increase in strength (NIBS 2019). The results indicate that the BCR tends to be higher in counties with a higher seismicity, and the national average BCR is approximately four considering a discount rate of 2.2 %. In addition, tenants obtained the greatest benefit, followed by title holders, communities, and lenders.

4.2. Benefits and costs of seismically retrofitting existing buildings

There are relatively many more studies that evaluate the economic value of mitigation strategies for existing buildings. However, as with new construction, the focus tends to be on bringing existing buildings "up to code" rather than exceeding current codes for enhanced performance. An exception is ATC (2009), which considers retrofit of wood-frame residential buildings for Immediate Occupancy, as discussed below.

Reinforced concrete buildings. Carofilis et al. (2020) investigate the retrofit strategies for school buildings constructed with reinforced concrete, precast concrete, and URM in Italy. The benefits of retrofit strategies are estimated using a combination of OpenSees, TreMuri, and PACT. The results indicate that adding steel braces or carbon fiber reinforced polymer (CFRP) strips to beam-column joints is not cost-effective for the reinforced concrete building but cost-efficient for the precast concrete building, with a payback period ranging from 56 to 83 years. The payback period for the URM school building is 32-39 years when CFRP strips are attached to both sides of masonry piers and spandrels. However, using CFRP strips and viscous dampers together is not economically feasible due to the high cost of dampers. Similarly, Haghpanah et al. (2017) evaluate three retrofit techniques for a concrete school building: base isolation, concrete jacketing, and steel jacketing. The results show that base isolation not only enables the building to meet the immediate occupancy performance criteria but also significantly reduces economic and casualty losses in the event of a large earthquake. On the other hand, the building with concrete or steel jacketing may be unoccupiable following a large earthquake. Smyth et al. (2004) analyze three retrofit methods for reinforced concrete apartment buildings. A five-story building in Caddebostan, Turkey, is selected to represent concrete buildings constructed under the 1967 code, which prescribes much smaller seismic loads than the current code. The benefit-cost analysis results suggest that the three retrofit methods are cost effective when the useful life of the building is longer than 10 years. Partial retrofit produces greater net benefits compared to full retrofit and bracing.

Wood-frame buildings. ATC (2009) compares three retrofit schemes for wood-frame buildings that contain soft stories, which are more flexible and weaker than the stories above due to lack of walls or frames (e.g., due to large openings for parking space).²⁰ Four representative residential buildings in San Francisco, CA, are selected to compute and compare the benefits (in terms of avoided loss to structures and contents) and costs of the retrofit schemes. The results show that the financial benefit of retrofitting

¹⁹ An importance factor of 1.5 is required for essential facilities by 2015 IBC, and lower values of 1.25 and 1 are used for designing buildings falling into Risk Category III and Risk Categories I and II, respectively (ASCE 7-16).

²⁰ One scheme uses steel cantilevered columns and greater shear walls to strengthen the soft story, which would allow residents to remain in their units after a major earthquake and strong aftershocks, analogous to retrofitting for an immediate occupancy (IO) performance objective (ATC 2009).

decreases as the size of an earthquake increases, as retrofitted buildings may be just as damaged as nonretrofitted buildings in an extreme earthquake. The direct construction costs for the four buildings, accounting for labor, equipment, and materials costs, range from \$9000 to \$19 000 per unit, while avoided losses range from \$24 000 to \$52 000 per unit. Similarly, Porter et al. (2006) evaluates retrofit strategies for wood-frame residential (single and multi-family) buildings in California, and finds that retrofit is only cost-effective in regions near fault or on soft soil.

URM buildings. Paxton et al. (2017), Goettel (2016), and Gibson et al. (2014) estimate the benefits and costs of retrofitting URM buildings at the city scale. Modifications are made to Hazus models to reflect local conditions and code requirements. Gibson et al. (2014) perform a benefit-cost analysis for URM buildings in Seattle, WA, and find that retrofit is not cost-effective because of high initial cost. However, some benefits that are not included in the analysis may weaken the conclusion, such as the increase in building's market value, reduction in building's insurance costs, greater historic preservation, and maintaining community visual character (Gibson et al. 2014). Similarly, Goettel (2016) conducts a benefit-cost analysis for URM buildings in Portland, OR, but finds that retrofitting URM buildings is cost-effective considering a 2 % discount rate and a 50-year useful life of the buildings. However, the results are subject to large uncertainty because the benefit-cost ratio is highly sensitive to discount rate and useful life. In addition, some benefits are not evaluated in this study, such as avoided damage in buildings adjacent to URM buildings, prevented fire following earthquake damage, prevented water damage from failed water pipes, decreased insurance costs, avoided injuries of visitors outside the buildings, and extended useful life of URM buildings (Goettel 2016). Paxton et al. (2017) analyze URM buildings in downtown Victoria, British Columbia, Canada. The benefit-cost results show that partial retrofit (adding tension anchorage to all floor-to-wall interfaces and adding out-out plan bracing to all slender walls) is cost-effective in enhancing the seismic performance of URM buildings, whereas full retrofit that complies with 2012 International Existing Building Code is not economically feasible even though it allows the least economic and casualty loss during earthquakes.

4.3. Benefits and costs of enhanced building designs in other domains

There is an extensive literature that analyzes the benefits and costs of meeting or exceeding building codes for other hazards, such as hurricane (Simmons et al. 2017) and tornado winds (Simmons et al. 2015). Unlike seismic events that cause damages to nearly all components of a building, wind damage is mainly focused on the envelope and roof-covering components, but hurricane winds followed by floods can result in water damage to building contents.²¹ Finally, Berry and Davidson (2016) present a review of methodologies for economic evaluation of energy-efficiency upgrades in more stringent codes, including benefits and costs not typically considered, with a case study in Australia.

²¹ At the intersection of earthquake and wind, Joyner and Sasani (2020) assess the performance of two seven-story concrete buildings that were built in compliance with ASCE/SEI 7-10 seismic and wind design requirements for Risk Category II.

5. The FR-BCA Framework

Based on a thorough review of the literature, we present the following three-step process to standardize benefit-cost analysis (BCA) for the evaluation of seismic risk reduction of buildings. This process is generic and can apply to both building codes for new construction as well as the retrofit of existing buildings. Where applicable, we point out differences in the process for each case. At each step of the process, we highlight the key components with respect to functional recovery, where there exist standard values for monetizing recovery-associated losses, and where gaps remain.

The objective of the functional recovery benefit-cost analysis framework (FR-BCA) is to tailor the process specifically for enhanced seismic design standards, rather than to reinvent the wheel with respect to conducting benefit-cost analysis. We outline a broad range of potential benefits and costs relevant to recovery-based design, with particular attention to impacted stakeholders and methods and tools for estimation. For many benefits, methods and tools may not be available and we highlight this gap as a research need.

The greatest challenge to conducting BCA for functional recovery is obtaining cost data. There is no current design standard for functional recovery on which to base construction costs.

Steps for conducting FR-BCA

Step 0: Set analysis parameters (r: real discount rate, T: planning horizon, m: mitigation option).

The choice of discount rate has been discussed thoroughly in the literature (e.g., Gibson et al. 2014). A typical discount rate is between 2 % and 10 %. The Office of Management and Budget (OMB) recommends 7 % (FEMA 2009). This may be critical if a project is federally funded or managed. In all other cases, the choice of discount rate is left to the discretion of the decision maker.

A typical planning horizon for built infrastructure such as buildings is the building's expected useful life. This tends to be somewhere between 50 and 75 years for new buildings (NIBS 2019), with T = 30 a reasonably accepted planning horizon for existing buildings after seismic retrofits (FEMA 2009).

Step 1: Estimate benefits, B_i, of action *i*.

In many benefit-cost analyses, this may be the most challenging step. For FR-BCA, this step requires first identifying any assets that are sensitive to earthquakes and then estimating the relationship between the severity of expected losses (damages) and the ground shaking hazard. The benefits of recovery-based design may then be estimated from the avoided losses under design option *i* relative to the status quo, as discussed in more detail in Sec. 5.1.

Step 2: Estimate costs, C_i, of action *i*.

In principle, estimating the cost associated with a particular design option is more straightforward than estimating the benefits. In practice, obtaining relevant cost information can also be challenging, especially for new design or retrofit standards for which there are no actual construction costs available. This is discussed in more detail in Sec. 5.2.

Step 3: Compare benefits and costs.

Once benefits and costs are estimated, the analyst can compare benefits to costs using two metrics:

Benefit-cost ratio (BCR): B / C

Net present value (NPV): B - C

where a BCR > 1 or NPV > 0 imply that the benefits of the design option outweigh the costs. Of course, these criteria may be sufficient but not necessary. That is, while a BCR > 1 should encourage adoption of the design option, a BCR < 1 need not discourage it. The BCA is limited to how well it can quantify the losses and, thus, the benefits. If key benefits are excluded (e.g., externalities or non-market values) or if loss estimation is flawed, then the BCR is likely to underestimate the relative benefits of the design option.²² Moreover, there may be other, intangible reasons (e.g., political will) for recommending adoption of a design option.

Step 4: Okay, we lied. There's a fourth step, but this step is optional.

Given estimates from steps 1 and 2, one can distribute benefits and costs across stakeholders (to obtain tiers of impacts). If we then compute the BCR for each stakeholder (or impact tier), we obtain a *distributed* BCR for each tier. Alternatively, once the BCR is computed, one can distribute the total BCR to obtain fractional BCRs for each tier to obtain BCR *shares per tier*. This step is rarely adopted, but there are exceptions (e.g., Cutfield and Ma 2015; NIBS 2019).

Moreover, sensitivity analysis can be used to examine whether the BCR shifts dramatically when inputs vary due to uncertainties in building's useful life, inflation rate, benefit and cost assumptions, hazard level, and model simulations. It is often helpful to determine the sensitivity range for each input and identify the inputs most important in estimating the baseline BCR (Gibson et al. 2014; Porter et al. 2006).

5.1. Estimating benefits associated with enhanced building design

As discussed above, the benefits from recovery-based design will come largely from avoided losses, which requires quantifying expected losses from an earthquake event. Of course, as we discuss below, there may also be other benefits that accrue independently of the occurrence of an earthquake (e.g., extending the useful life of a building).

Expected losses are quantified in annual terms (Expected Annualized Losses, or EAL), with (EAL_i) and without (EAL_0) the action. A general formula for EAL is:

$$EAL = \int_0^\infty L|dP(L)| \tag{4}$$

where P(L) is the PEER mean annual rate of exceedance for the loss L (Porter 2003; Krawinkler et al. 2006; Mitrani-Reiser 2007):

²² Note that one can also subtract the "salvage value" of the building from the cost (e.g., Kappos and Dimitrakopoulos 2008).

$$P(L) = \int_{dm} \int_{edp} \int_{im} G(L|dm) |dG(dm|edp)| |dG(edp|im)| |dP(im)|$$
(5)

where $G(x|y) = 1 - F(x|y) = P(X \ge x|Y = y)$ denotes the "exceedance" probability (i.e., the complementary CDF, or survival function); *dm* denotes the *damage measure* (e.g., damage state); *edp* denotes the *engineering demand parameters* (e.g., max drift); and *im* is the *intensity measure* (e.g., pga). P(im) is expected rate of return of the ground shaking hazard derived using a Poisson model for the annual rate of exceedance for the intensity measure *im*. Colloquially, P(im) is the "hazard curve."

In the literature on performance-based earthquake engineering (PBEE), P(L|dm) is also written as P(dv|dm) where dv is a *decision variable*, such as damage, downtime, and casualties (Mitrani-Reiser 2007).²³

If we use a discrete damage measure such as damage state (DS) then EAL may be computed as:

$$EAL = \int_{edp} \int_{im} \sum_{DS} L_{DS} \times G(L|DS) d(DS|edp) dG(edp|im) dP(im)$$
(6)

The G(L|dm) is typically calculated using consequence functions that link damage to losses (typically the loss ratio, which is the ratio of repair costs to replacement costs). Consequence functions may be defined at the global-level (e.g., building level), for a coarse level of detail, or at the component-level, for finer level of detail. Consequence functions are typically derived from data on actual losses or from expert judgement (Odabasi et al. 2020). Hazus, for instance, provides detailed global-level consequence functions for various occupancy types and damage states. On the other hand, FEMA P-58 provides component-level consequences for global assessment, component-level performance must be aggregated into global-level consequences considering the interdependencies in the repair of multiple components within a given asset (FEMA 2018, Cook et al. 2022).

The benefit of risk reduction (new construction or existing building) is the present value of the avoided losses due to action *i*:

$$B = [EAL_0 - EAL_i] \sum_{t=1}^{T} (1+r)^{-t}$$
(7)

Equivalently, by sum of geometric sequences

$$B = [EAL_0 - EAL_i] \times \left[\frac{1 - (1 + r)^{-T}}{r}\right]$$
(8)

²³ There is also a literature that explores simplified methods for computing EAL (e.g., Porter et al. 2004; Solberg et al. 2008; Ramirez et al. 2012; Welch et al. 2014; Cardone et al. 2017; Cardone et al. 2019a, 2019b).

Thus, the key to benefit calculation is to identify and quantify the relevant potential losses, since avoided losses constitute the major portion of benefits.

Remark: Note that the preceding discussion focuses on quantifying a point estimate of the benefit. In principle, one can also compute the variance of benefits by calculating the variance of avoided losses: $Var(B) = V[Avoided \ Losses] \sum_{t=1}^{T} (1+r)^{-2t}$ where

$$V[Avoided \ Losses] = V[EAL_0] + V[EAL_i] - 2Cov[EAL_0, EAL_i]$$
(9)

Recall that $V[X] = E[X^2] - E[X]^2$ and Cov[X,Y] = E[XY] - E[X]E[Y]. Thus, to calculate variance of avoided losses, we need to calculate $E[EAL_i^2]$ for i = 0, 1 and $E[EAL_0EAL_i]$, which can be computed using the same formula; e.g., $EAL^2 = \int_0^\infty L^2 |dP(L)|$. This may be valuable for uncertainty quantification.

Table 5.1 summarizes the potential benefits relevant to recovery-based design, including the affected stakeholders and typical methods for monetizing those benefits, where available.²⁴ As the preceding discussion illustrates, benefits are obtained by estimating avoided losses with respect to a recovery-based design option. Thus, the benefits in Table 5.1 are presented in terms of the relevant losses that should be estimated to obtain avoided losses. The key to estimating benefits for functional recovery is to estimate downtime (time to re-occupancy or recovery of function), as discussed in more detail in Sec. 5.3. Given estimates of downtime, one can estimate monetary losses associated with downtime using Table 5.1. In addition, there may be co-benefits that accrue even in the absence of an earthquake during the planning horizon, including reduced environmental impacts (Belleri and Marini 2016; Dong and Frangopol 2016), extended lifespan of the asset (Di Bari et al. 2020; Belleri and Marini 2016), and increased economic resilience (Fung et al. 2021a). Since such co-benefits are not dependent on enhanced performance, and moreover established methods for estimating them are not available, we do not cover them in detail but note that they may be critical to making a business case for recovery-based design.

Note that in practice, it may not be feasible to calculate the full range of benefits. Some benefits, such as impacts on quality of life and property value, tend to be highly idiosyncratic and are difficult to monetize, especially if there is no readily available data on the values of such benefits. Other benefits, such as avoided damages to and from neighboring buildings, require more sophisticated models for loss assessment that incorporate multiple assets and their spatial dependence. As a result, Table 5.1 also identifies additional needs for tools and data required to monetize these benefits. Each category of losses and the associated potential benefits are discussed in the subsections that follow.

²⁴ For a review of methods and tools available, see the review of Zhang et al. (2022).

Table 5.1. Potential direct and indirect loss categories and the associated benefits from enhanced design, as well as the potential beneficiaries and
methods for monetizing.

	Loss	Benefit	Beneficiaries	How to monetize	
Direct	Building damage	Avoided repairs to structural and nonstructural components	Owner, Taxpayers (government assistance)	<pre>(Repair cost × Prob[Repair]) + (Demolition cost × Prob[Demolition]) OR x % of Replacement cost (x = Damage ratio)</pre>	
	Contents damage	Avoided loss of contents (e.g., equipment, furniture)	Owner ⁺ , occupants	x % of Replacement cost ($x = Damage ratio$)	
	Casualties (deaths and injuries)	Avoided deaths and injuries	Owner ⁺ , occupants, visitors	(VSL) \times (number of deaths/injuries)	
Indirect	Economic losses	Business interruption			
		Avoided loss of business income	Occupants, community (local, national) due to direct and indirect impacts	(Average daily revenue) × (days of business interruption)	
		Avoided loss of rental income	Owner	(Average rental rate/sq m) × (Fraction of area rented) × (Days of business interruption)	

	-	
Avoided loss of productivity	Occupants, community	See literature on section 5.1.2.*
Avoided reduction in customers	Occupants, community	See literature on section 5.1.2.**
Avoided reduction in employment	Occupants, community	See literature on section 5.1.2.**
Insurance-related losses		
Avoided increase in insurance premium costs	Owner, occupants	(Premium after earthquake event) - (Premium before earthquake event)*
Avoided delays in settling insurance claims	Owner, occupants	See literature on section 5.1.3.**
Improved equity through reduction in regressive hazard insurance	Owner, occupants	See literature on section 5.1.3.**
Indirect property losses		
Avoided property value decrease (due to physical damages, capitalized risk)	Owner, owners of similar properties	See literature on section 5.1.4.*
Avoided reduction in tax base	Community	See literature on section 5.1.4.*
Avoided delays in recouping return on	Owner, Developer	See literature on section 5.1.4.***

	investment due to delays		
	Supply chain disruption		
	Avoided supply chain delays (upstream and downstream)	Owner ⁺ , occupants ⁺ , upstream/ downstream partners	See literature on section 5.1.5.*
Social losses	Avoided displacement costs	Owner ⁺ , occupants, taxpayer ⁺	See literature on section 5.1.6.
	Avoided loss of life quality due to deterioration in physical or mental health	Owner ⁺ , occupants, neighbors	See literature on section 5.1.6.
	Preserve culture/history/character and help attract tourism investments	Community (local, national)	See literature on section 5.1.6.
	Avoided loss of sense/connectedness/comm unity	Local community	See literature on section 5.1.6.**
	Reduction in perception of risk due to improved performance	Owner, occupants, owners of similar properties, local community	See literature on section 5.1.6.***
	Reduced burden on emergency response	First responders	(Number of first responders at risk) × (QoL × WTP) OR

				(Number of injuries) × (Medical expense)
		Reduced crime rate	Community	See literature on section 5.1.6.**
		Impacts on underserved communities/reduction in affordable housing	Community	See literature on section 5.1.6.**
		Government assistance	Taxpayer	See literature on section 5.1.6.**
	Physical losses	Avoided clean-up costs and closures due to on-site/off- site debris	Owner, occupants, visitors	(Clean-up costs) × (Amount of debris) NB: Do not double count. This is used to attribute the fraction of losses due to debris.
		Avoided energy consumption, greenhouse gases emissions, environmental pollution due to repairs/demolition	Owner, community (local and at large)	See literature on section 5.1.7.
4 D 11	TT'/1 1 11			

Notes: Building owner = Title holder. VSL = Value of a Statistical Life. WTP = Willingness-to-pay. Prob = Probability. QoL = Quality of Life.

⁺ Potential impact.

* Research is needed with precedent in related literature. ** Research is needed without precedent.

*** Highly speculative.

5.1.1. Direct benefits

Building and contents damage are standard outputs of a performance-based earthquake risk assessment (e.g., FEMA P-58). Note that these losses are typically calculated based on the replacement value of the building (the damage ratio). Casualties (deaths and injuries) may not be standard outputs, but methods exist for estimating casualties based on building performance (e.g., using the PEER framework in Eq. (2)); see for example Mitrani-Reiser 2007. Note that for recovery-based design, avoided casualties may be low relative to code-conforming design, but are expected to be higher relative to older buildings (that is, the benefits from reduced casualties are expected to be much higher for retrofits than for new construction).

5.1.2. Indirect economic benefits

The next category of losses in Table 5.1 covers indirect losses due to business interruption. A reasonable approximation to the number of days of business interruption is days to recovery of function (though it may be shorter conditional on the ability to rent temporary space and resume operations). Assuming this can be estimated (for instance, using the methods discussed in Sec. 5.2), indirect economic losses are calculated as:²⁵

$$Business \ loss = (loss \ per \ day) \times (days \ of \ business \ interruption)$$
(10)

The loss per day may be relatively easier to estimate for some benefits than for others. For instance, business income is typically calculated based on average daily revenue. Absent data on actual revenue, daily revenue may be estimated from economic measures such as proprietor income or value added.²⁶ Similarly, rental income may be based on market rental rates, which vary by building occupancy.²⁷

On the other hand, loss of productivity may be more challenging. To estimate loss of productivity, we need to estimate both productivity and the impact of business interruption on productivity:

 $Productivity \ loss = (Productivity \ per \ day) \times (Productivity \ decrease) \times (days)$ (11)

Productivity is typically measured as wages or gross value added (Seppanen et al. 2004). The more challenging step is estimating the impact of business interruption on productivity, which can be measured directly through surveys or indirectly through observational studies. For example, the literature that studies the impact of building environment (such as temperature and indoor air quality) on worker productivity relies on a combination of thermal comfort models, questionnaires or surveys to elicit subjective impacts, and indirect assessment of occupant performance through measures such as absenteeism, turnover, and grievances (Horr et al. 2016). Using models from the literature, Kershaw and Lash (2013) estimate potential productivity losses from climate change for office workers in England between \$23 and \$31 per sq ft (\$248 and \$334 per sq m) annually. For enhanced seismic codes, we are

²⁵ For more accuracy, it is typical to use income per square foot per day, then multiply that by the building in the study's square footage.
²⁶ The Bureau of Economic Analysis (BEA) defines proprietor income as the "current-production income of sole proprietorships, partnerships,

and tax-exempt cooperatives" that excludes dividends, rental income, and interest payments (BEA 2018a). Value added is defined as the "gross output of an industry or a sector" net of intermediate purchases (BEA 2018b).

²⁷ Market rental rates, as well as estimates of the fraction of a building that is rented, can be obtained from Hazus (FEMA 2012a) or BOMA (2018).

not aware of literature that estimates impacts on productivity.²⁸ Thus, this is an identified research gap.^{29,30}

Finally, reductions in customers and employees are more difficult to quantify. The data available in the literature is anecdotal at best (Potter et al. 2015). We caution that while estimating such impacts may be valuable depending on the application, there is a danger of double counting benefits.³¹ The potential losses in Table 5.1 assume the building has restored some level of functionality, but nevertheless suffers a reduction in customers or employees. This could be due to cascading effects from the event that result in dislocation, emotional well-being, or increased risk perception due to observable damage *even if the building is functional*. Losses due to reduction in customers could be estimated as

 $Customer \ loss = (Revenue \ per \ customer) \times (Number \ of \ lost \ customers)$ (12)

Similarly, losses due to a reduction in employees could be estimated as

$$Employee \ loss = (Revenue \ per \ customer) \times (Number \ of \ lost \ customers)$$
(13)

Note that neither of these losses are directly tied to loss of functionality. However, restoration of function may reduce these impacts relative to code-conforming or existing buildings. Estimating the numbers of lost workers or customers may require building this into performance assessments, surveys, or potentially economic impact assessment models for which impacts can be downscaled (Sarmiento 2007; Belasen and Polachek 2009). We leave this as an open area for research.

5.1.3. Insurance-related losses

The insurance-related losses in Table 5.1 are distinct from indirect economic losses as they are intended to capture impacts on insurance markets, which may be difficult to quantify. In principle, estimating the (avoided) increase in insurance premiums following an earthquake should be straightforward. In practice, it requires access to proprietary information from insurance companies. If a building owner, manager, or developer, or a community, has experienced an increase in insurance premiums following an event, they may use their knowledge as an approximation.

Delays in settling insurance claims result in delayed repairs. Potter et al. (2015) note delays of two years or more following the Canterbury earthquake sequence. Such delays, as well as delays due to processing loans, post-earthquake inspection, and contractor mobilization are called impeding factors in the literature on downtime (Almfuti and Wilford 2013). REDi (2013) characterizes financing delays as the most uncertain impeding factor, with the caveat that insurance deductibles often exceed the expected losses. The recommended approach to mitigate such losses is to limit expected losses so that repairs can be financed with available funds. Thus, while the avoided losses may be substantial, the expected impact of

²⁸ The California Governor's Office of Emergency Services (Cal OES) suggests using \$8,736 per worker for benefit-cost analyses of seismic retrofits for residential buildings. However, there are no details on how this figure was obtained. https://www.caloes.ca.gov/RecoverySite/Documents/2019%20BCA%20Presentation.pptx (Accessed: 2021-08-20).

²⁹ Another example uses cognitive performance scores to assess the impact of optimizing window views and daylight on productivity (MacNaugthon et al. 2021).

³⁰ The literature on productivity losses due to the COVID-19 pandemic may be able to provide insight on measurement approaches.

³¹ For example, loss of business income will be directly tied to loss of customers and employees. One could think of these losses as ways to disaggregate loss of business income, if that is of interest and the data is available.

impeding factors on delays will vary case by case.³² As noted in Sec. 5.2, this is an active area of research as it is critical for estimating downtime. In principle, if a performance assessment can estimate downtime due to repairs ("rational downtime") as well as downtime due to impeding factors ("irrational downtime"), we can apply similar logic as in Sec. 5.1.2 to disaggregate business losses (Comerio 2006).

On the other hand, losses due to reduced equity are more speculative but a potential research need. Owen and Noy (2017) examine half a million claims for the New Zealand earthquake insurance program and find that the system is regressive; that is, lower income households pay proportionately more than higher income households. This is largely due to the nature of the premium, which is risk-based and thus tied to socioeconomic status. Thus, in principle a reduction in insurance premiums or coverage could improve equity by reducing the disproportionate burden on lower income households. However, as Owen and Noy (2017) note, there has been little research to date on natural hazard insurance programs and their impact on equity. Moreover, to go one step further and monetize the benefit would require estimating the burden of such regressive insurance schemes on different kinds of households. In part, this would require data on insurance premiums, but it would also require socio-economic data such as household income. Thus, this is left as an open area for research.

5.1.4. Indirect property losses

In addition to the direct losses from building and contents damage, there may be indirect losses associated with property damage. For instance, a damaged building that survives an earthquake event may suffer from a reduction in property value due to increased salience of perceived risk (Timar et al. 2018a).³³ In addition, buildings that provide enhanced performance may experience a property price premium (Filippova et al. 2017).

Such benefits for property value are typically estimated using willingness-to-pay (WTP) methods that attempt to capture the capitalization of risk into property prices. D'Alpaos and Bragolusi (2020) use stated preference methods (contingent valuation) to estimate a positive price premium for earthquake risk reduction in the Italian housing market. Filippova et al. (2018) use revealed preference methods (hedonic regression) to estimate the impact of earthquake risk after the Canterbury Earthquake Sequence on commercial buildings in Auckland, New Zealand and find a 12.5% price reduction for office buildings (and no effect on retail). Using fault map changes in California over the period 1970-2010, Singh (2016) finds that, on average, home prices increase by 1.8 % for every one-mile increase from the fault zone.

Moreover, property value losses may have cascading impacts for the community by reducing the tax base. Estimating the benefit to the community, at least from the perspective of a single asset, is not straightforward. First, the marginal impact of a single building on the tax base may not be substantial enough to include in an economic analysis (and is likely to vary highly across buildings). Moreover, meaningful reductions in the tax base are more accurately captured as general equilibrium impacts due to other factors such as displacement of businesses and households. Thus, the most appropriate method for estimating impacts on the tax base is likely economic impact analysis. However, since economic impact analysis measures regional impacts, the use of such methods would require assumptions for downscaling

³² Relying on private or federally-backed loans may result in much higher delays. Table 8 in REDi (2013) provides median and dispersion estimates for different kinds of impeding factors.

³³ Several papers find that risk disclosure alone does not affect property values, but that earthquake events make such disclosures more salient. See for instance Timar et al. (2018b), Huang (2021). Moreover, salience in one region may increase in response to earthquakes in other regions, even if unaffected (Fekrazad 2019).
impacts with respect to a single building. While this is possible, we are not aware of applications in the literature and thus leave this as a research need.

Finally, owners and developers may experience deferred recouping of investment costs if repairs and impeding factors cause massive delays. However, this loss is speculative, and we are not aware of any methods in the literature for monetizing this benefit.

5.1.5. Supply-chain disruption

The position of businesses in modern supply chains means that local disruptions can have global impacts. Recent research suggests that disruptions due to local hazards can propagate through the supply chain to create losses that exceed local impacts (Thomas and Helgeson 2021). A building's business interruption may have indirect effects on its supply chain, both in terms of reduced purchases or delayed payments upstream (to sellers) or delayed delivery downstream (to buyers). For instance, in a survey of Japanese firms affected by the 2011 floods of Thailand, Ye and Abe (2012) find that while only 19 % of respondents experienced direct damage, 78 % experienced indirect losses due to supply chain disruption. Thus, from the perspective of a building that has been designed for functional recovery, the relevant avoided losses are to its upstream and downstream supply chain partners. If a building is functional but cannot obtain products due to upstream disruptions, there is nothing enhanced design can do for that building's occupants.³⁴

Despite the abundance of research on both supply chain disruptions and business interruption, there is relatively little literature that estimates firm-level losses due to upstream or downstream propagation of a disruption (Katsaliaki et al. 2021). Dormady et al. (2022) conduct a survey of businesses impacted by Hurricanes Sandy and Harvey and find that supply chain disruptions account for 23 % of business interruption losses, on average. In a review of the empirical literature on supply chain disruptions, Katsaliaki et al. (2021) find that supply chain disruptions lead to an average drop in profitability of 107 %, a drop of 7 % in sales growth, and an increase of 11 % in costs. Dolgui et al. (2018) review the literature on modeling and quantifying the propagation of supply chain disruptions and present a few case studies of large-scale disruptions that do not generalize to typical buildings. Nevertheless, we observe a research need in estimating willingness-to-pay to avoid supply chain disruptions.

Due to the complexity of modern supply chains, economic impact analysis methods are well suited to capturing upstream and downstream losses due to local disruptions, and the cascading impacts of those losses locally (the feedback loop).³⁵ Both Input-Output (I-O) and agent-based models (ABM) can be used to estimate upstream or downstream impacts of business interruption on its supply chain. For instance, Henriet et al. (2012) disaggregate I-O tables to represent a regional economy as a network of individual production units. However, while this approach captures more complex relationships among a production network, it is not intended to capture firm-level impacts of supply chain disruptions. Inoue and Todo (2019) use an ABM to model the propagation of impacts of the 2011 Great East Japan earthquake through the supply chain and find that the indirect effects due to propagation are 10.6 % of GDP, while the direct

³⁴ Wagner and Bode (2006) survey German executives and identify global sourcing as a key driver of catastrophic supply chain risk (e.g., due natural hazards, terrorism, political instability). Surprisingly, dependence on a few suppliers is correlated with reduced catastrophic risk.
³⁵ Dolgui et al. (2018) and Katsialiaki et al. (2021) review methods for modeling supply chain disruption, including numerical optimization and simulation methods. In principle, these can be combined with economic impact models to estimate losses from supply chain disruptions.

effects are only 0.5 % of GDP.³⁶

In principle, it should be possible (with additional assumptions) to downscale estimates from those models to a single building. However, while there is plenty of research using I-O or ABM methods to estimate the regional impacts of a supply chain disruption, we are not aware of any research that downscales results to a single building. Thus, this is left as speculative and a potential area for further research.³⁷

5.1.6. Indirect social losses

In addition to the economic impacts discussed above, loss of occupancy and building functionality impacts people in other less tangible ways. We call such impacts indirect social losses. In this subsection, we discuss some key categories of indirect social impacts as well as methods for monetizing social losses (predominantly willingness-to-pay methods).

Displacement. Loss of occupancy and function can result in displacement of building occupants. Displacement of occupants is associated with a range of measurable costs, including moving and travel costs (e.g., rental costs, transportation costs, transportation time) to households and businesses. Some displacement costs, such as the cost of travel to a new location, can be considered fixed or one-time costs, while other displacement costs, such as rental costs for temporary space, can be considered recurring costs. Thus, fixed costs are a function of *whether* a building loses occupancy/functionality, while recurring costs are a function of *how long* a building loses occupancy/functionality. Thus, losses associated with displacement may be estimated as

 $Displacement \ loss = [DC_{fixed} + (DC_{recurring} \times days)] \times (Number \ of \ displaced \ persons)$ (14)

where DC_k are displacement costs for $k = \{fixed, recurring\}$ and *days* is the number of days until restoration of occupancy/functionality. Estimates of DC_k may be obtained using stated or revealed preference methods. Whitehead (2003) uses a combination of stated and revealed preference methods to estimate the opportunity costs associated with hurricane evacuation costs. Average total recurring costs during displacement (including lodging, food, and entertainment) are estimated between \$20 and \$275, while average total fixed costs (including travel and the value of time) are estimated between \$58 and \$195.³⁸

In addition, population displacement, either temporary or permanent, can have a significant effect on a city or regional economy after a hazard event. Oftentimes, these numbers are difficult to calculate because official sources of information typically come from census data, which are collected infrequently. For example, using 2013 data New Zealand reported a 2 % decrease in Christchurch's city population after the 2011 earthquake sequence, but a 3.4 % regional population increase over the same time period. However, the calculations are dependent on 2006 census data to provide a baseline (Bayer 2013). Newer

³⁶ Okuyama (2007) presents a review of I-O and related models for economic disaster impact assessment. Modern approaches tend to combine I-O tables and ABMs (e.g., Hallegate 2019).

³⁷ Recently, there has been a lot of research into the supply chain impacts of COVID-19 non-pharmaceutical interventions such as lockdowns (e.g., Guan et al. 2020). This literature could be useful as well.

³⁸ Note that recurring costs are estimated over the total time displaced. The estimation of expected costs is conditioned on storm severity and evacuation model (no order, voluntary, mandatory) rather than length of evacuation (Whitehead 2003). Nevertheless, the same methods could be applied to the estimation of daily displacement costs. Reported costs are in 2003 dollars.

methods, utilizing social media tracing or cellular phone location data, may prove to provide more accurate and timely estimates for population displacement and outmigration (Acosta et al. 2020; Yabe et al. 2020). It is also important to note that minority and disadvantaged populations are oftentimes disproportionately burdened via displacement and recovery (Phillip 2015).

Government assistance. Following a disaster, local building and safety departments organize teams of inspectors to identify damaged buildings. Inspectors will conduct a safety assessment and issue a colored tag on the inspected structure, with "green" indicating no hazards, "yellow" indicating the building is moderately damaged and its habitability is limited, and "red" indicating the building is unsafe and should not be entered under any circumstances (Eguchi et al. 1998). Inspectors may also provide rough estimates for the repair costs of inspected buildings. Eguchi et al. (1998) report that more than 105 000 inspections were carried out by local governments following the Northridge Earthquake.

Government funds are important to post-disaster building repairs and displacement of affected residents. Following the 1994 Northridge Earthquake, a total \$13 billion federal fund was allocated to aid earthquake victims (Petak and Elahi 2001). About \$1.424 billion was used to provide temporary housing, emergency home repairs, and mortgage assistance, and \$0.167 billion was used to assist in personal property replacement, permanent repairs, and transportation, medical, and funeral expenses (Petak and Elahi 2001). In addition, \$4.6 billion was designated to repair and replacement of damaged infrastructure, emergency service, and debris removal (Petak and Elahi 2001).

Burden on taxpayers. There are indirect losses to taxpayers via direct disaster assistance and other forms of financial support. Stein and Van Dam (2019) report that, since 1990 in the United States, federal spending for disaster relief appropriations has increased eight-fold. They also highlight a Pew Charitable Trust report that states that only about 44 % of total disaster spending comes from this disaster relief fund, with other amounts coming from other public and private entities. The Congressional Research Service (CRS 2021) identifies \$12 billion in total gross disaster relief fund appropriations for fiscal year 2019. The Internal Revenue Service (IRS 2021) data implies that 154 million individual taxpayers each pay at least \$80 per year for disaster relief. Moreover, Deryugina (2017) argues that natural hazard events also lead to non-disaster government funding, such as unemployment insurance and public medical payments, which actually exceeds the value of direct disaster aid. Therefore, taxpayer burden for disaster and disaster adjacent costs is underestimated, particularly as reflected only by federal disaster relief fund expenditures. In recent years, hazard mitigation has also become an important component of disaster spending. In 2018, Congress incentivized local mitigation measures by increasing the federal share available for disaster recovery in states that have invested in mitigation (Pew 2018). Functional recovery design would enable cost reductions in all areas of taxpayer burden including disaster response, mitigation, and other related costs.

Deterioration of physical or mental health. Loss of occupancy/functionality and the resulting displacement may have compounding negative impacts for both individual physical and mental health. Hogg et al. (2016) study the impact of different modes of displacement on mood and anxiety following the Canterbury earthquakes. They find that temporary relocation is a short-term risk factor. Moving within the city mitigates impacts on mood/anxiety but returning is a significant risk factor. Out-of-city movers are especially vulnerable. Fussell and Lowe (2014) analyze pre- and post-Katrina survey data on black single mothers and find that displacement increases general psychological distress and perceived stress. However, physical and mental health impacts of post-disaster displacement are less studied.

Uscher-Pines (2008) finds only four of 24 articles that assess displacement impacts focus on physical health. Moreover, the article reveals "weak study designs, inconsistent results, and inattention to physical health impacts and the challenges facing vulnerable populations." In addition, definitions of relocation/displacement are not consistent (NB this article is from 2008).

Disasters also affect the mental health of a broad population. Bromet et al. (2016) estimate that the incidence of post-traumatic stress disorder (PTSD) following natural and human-made disasters is anywhere from 3-5 % to 20-40 %. Morris and Deterding (2016) suggest that post-Katrina dispersion of social networks is associated with PTSD. There is "a lack of deep belonging and a lack of mattering as they are unable to fulfill obligations to important distant ties" due to increased physical distance from social network members.

Physical and mental health are non-market goods, meaning there is no objective market price to value negative impacts on health (Sidney et al. 2017). In the public health literature, the standard approach for estimating impacts on health rely on monetizing quality-of-life (QoL) adjustments due to a change in health outcomes (Lachaine et al. 2012). This requires both (1) a way to measure changes in QoL and (2) a way to monetize the value of changes in QoL.

$$Loss of QoL = (Change in QoL) x (Value of QoL)$$
(15)

Two approaches can be employed to measure the *change in QoL*. One approach is to estimate reductions in health care costs due to an intervention (Sidney et al. 2017). Another approach relies on questionnaires that are developed for specific health interventions to assess impacts on QoL or quality- and *quantity*-of-life, known as quality-adjusted life-years, or QALYs (Whitehead and Ali 2010). As an example, the European Organisation for Research and Treatment of Cancer (EORTC) has developed a widely used questionnaire to assess QoL for cancer patients: the EORTC QLQ-C30 (Fayers et al. 2002). Similar questionnaires exist for mental health interventions (Connell et al. 2014).

Given a measure of QoL, the next step is to estimate willingness-to-pay (e.g., Huang et al. 2018) or potential savings (e.g., Sidney et al. 2017) from a particular QoL change. For instance, Sidney et al. (2017) study the impact of employer well-being programs on employee health. Using a combination of medical and pharmaceutical claims, the authors estimate a total of \$3060 saved in annual costs per reduction in disease occurrence.³⁹ Moreover, employees with disease occurrence save \$62 annually on average, while employees without disease occurrence save \$26 annually on average.⁴⁰

Burden on first responders. The ability of a building to recover function quickly after an earthquake should reduce the need for an extended emergency response presence. The standard approach to estimating the burden of disasters on first responders⁴¹ is to measure the impacts on their physical and

³⁹ It is worth noting that Sidney et al. (2017) is a notable example of a study that uses a combination of micro-level employer data with inputoutput data to estimate both direct and indirect benefits of an action. The study's methods could be adapted to preceding discussions such as supply chain disruption.

⁴⁰ Similar methods could be applied to estimate the value of QoL improvements due to improved building functionality; however, in the absence of retrospective data as in Sidney et al. (2017), such research would rely on stated preference methods.

⁴¹ Wanner and Loyd (2021) describe the command structure of response: the Incident command system (ICS). The first level is local response, including Emergency Medical Service (EMS), Medical Reserve Corps (MRC), and Community Emergency Response Teams (CERT), which are the immediate responders. If local capabilities are overwhelmed, state and regional response is available, including National Guard, EMS districts, and Nongovernmental Organizations (NGOs). Large-scale disasters may require federal response via presidential disaster declaration (PDD), which makes additional medical resources available.

mental health (Benedek et al. 2007). Thus, the methods discussed above can be applied specifically for measuring the value of QoL impacts from disaster response. To estimate the losses associated with loss of function, we also need to determine how many first responders would be impacted. For instance, if we are interested in mental health impacts,⁴² we might consider incidence of PTSD among first responders:

Mental health impacts

$$= (Rate of PTSD) \times (Number of first responders) \times (days)$$
(16)

In a review of the literature on wildland-interface fire, Thomas et al. (2017) report incidence rates of PTSD for firefighters between 13 % and 20 %. However, mental health impacts may increase over time. Berninger et al. (2007) find that one year and four years after the World Trade Center (WTC) 9/11 attack, the prevalence of PTSD among 10 074 firefighters increased from 9.6 % to 10.8 %. Wisnivesky et al. (2011) also note that one year and nine years after the WTC attack, the prevalence of PTSD increased from 12.8 % to 31.9 % for 27 449 rescue and recovery workers, including firefighters.

The number of first responders may vary by the level of emergency, particularly the number of people threatened, the type and size of the building, and the stage of the fire when firefighters arrive. Fire departments will designate an alarm (the 1st, 2nd, 3rd, etc. alarm) for major incidents, and the type of alarm determines how many firefighters are sent to the emergency (IFD 2021). As for the post-disaster emergency response, the National Urban Search and Rescue (US&R) Response System, established by FEMA in 1989, has 28 US&R task forces. Each task force is composed of 70 members specializing in search, rescue, medicine, hazardous materials, and logistics and planning (FEMA 2022). In addition, there were 1150 local US&R response teams by 2006, varying from one team per state to 79 teams in California (Denver et al. 2007). Thus, it is difficult to estimate the average number of first responders for an earthquake. However, it may be possible to estimate the number for specific scenarios.

Estimated mental health impacts can then be combined with monetized QoL measurements for reductions in occurrence in PTSD to measure the burden on first responders:

$$Burden on first responders = (Mental health impacts) \times (Value of QoL)$$
(17)

As an example, Butry et al. (2019) use national databases of fire incident reporting to estimate the economic burden of firefighter deaths and injuries and find the annual cost per firefighter to be between \$1468 and \$5412. Thomas et al. (2017) report an average of \$4075 in costs to treat PTSD among military personnel. We are not aware of any literature that explicitly considers the burden of earthquakes on first responders, but there is sufficient precedence in related literature that this is a promising topic for research.

Loss of cultural heritage. In addition to building damage and property value impacts, property losses can also result in losses of a community's cultural heritage. We use the term heritage building to denote a building that has historic or cultural value. Heritage buildings, which are typically protected by local building regulations, may have intangible value to the community that can be irreversibly lost after an

⁴² We use PTSD to illustrate potential impacts. Other potential mental health risks to first responders include suicide, substance abuse, and depression or anxiety (Fitzpatrick 2020). In addition, first responders face physical risk of injury and death.

earthquake.⁴³ While it is costly to retrofit heritage buildings that require preservation of character, the potential value of cultural losses may be much larger and potentially extend beyond the building itself. Of course, it should be noted that this benefit is only relevant for existing buildings.

Willingness-to-pay methods are best suited to estimate the value of preserving a heritage building's character. Lazrak et al. (2014) use spatial hedonic methods to estimate the value of cultural heritage in Dutch real estate markets. They find that buyers are willing to pay 27 % more for a building with a historic designation. Similarly, Andersson et al. (2019) estimate a price premium between 36 % and 60 % for historic buildings in Sweden using hedonic methods. Moreover, hedonic methods can be used to estimate the cultural externality from being located near a historic building. For instance, Lazrak et al. (2014) estimate a 0.28 % price premium for buildings within a 50-m radius of a historic building. Andersson et al. (2019) estimate the cultural externality value to be around 1 %. On the other hand, while Ahlfeldt and Maennig (2010) do not find a statistically significant premium for heritage designation in Berlin, they do find a cultural externality of up to 2.8 % for properties within a 600-m radius of a heritage building.

In addition to revealed preference methods, stated preference methods may also be useful for estimating WTP. Powell et al. (2015) survey building owners in Wellington, New Zealand following the Canterbury earthquake sequence as well as legal changes requiring seismic strengthening for earthquake-prone buildings. Respondents report an average increase in property value of 282 % for heritage buildings, in comparison to a 23 % increase for modern buildings. Moreover, stated preference methods may be used to supplement revealed preference methods. Alberini and Longo (2006) combine the travel cost method with contingent valuation to estimate a use value of between \$28 and \$44 per person for cultural monuments in Armenia, which imply that preservation policies can improve social welfare from \$ 2.8 million to \$4.2 million.⁴⁵

Loss of social cohesion. The relationship between a disaster and social cohesion can go in two directions. Social cohesion can improve community resilience to natural disasters. Ludin et al. (2018) find strong associations between community resilience and social cohesion among six flood-prone communities in Malaysia. Townshend et al. (2015) also find a "consistent significant positive correlation between cohesion and resilience" among rural Canadian communities experiencing disasters and evacuations. In addition, Thornley et al. (2015) indicates that social cohesion improved recovery after Canterbury earthquakes, though the existing hardships were exacerbated by the earthquakes. Hikichi et al. (2016) suggests that pre-disaster social cohesion is associated with lower rates of PTSD among survivors of 2011 Tohoku earthquake. Greene et al. (2015) notes that social cohesion increases resilience and reduces poor mental health outcomes in flood-affected areas of England.

On the other hand, natural disasters may enhance social cohesion. Calo-Blanco et al. (2017) suggests that social cohesion increases after earthquakes, though the effect erodes over time, especially for less severe events. Shigemoto and Kawachi (2020) find that there was no significant association between social cohesion and QoL in the short run following the 2008 Hurricane Ike, but a significant *positive* correlation

⁴³ Forte et al. (2021) provide a characterization of the characteristics that give cultural assets their value, as well as a review of methods for estimating lost value due to damage from an earthquake.

⁴⁴ Historic preservation can also have negative impacts due to constraints on development. For instance, Bade et al. (2020) study the price effects of being located within an area with a heritage designation in Auckland, New Zealand. They find that there is a 9.6 % penalty for being located within such an area, though they still find a positive cultural externality of 1.7 % on average for homes within 50-m of a heritage area building. ⁴⁵ All values in 2006 U.S. Dollars.

appeared in the long run (i.e., 15 months after the event).

In addition, the impacts on social cohesion have indirect effects on social behavior. Disasters may reduce crime due to prosocial behavior (based on the theory of the therapeutic community) or increase crime due to loss of social cohesion (based on the theory of social disorganization or the theory of routine activities).⁴⁶ Prelog (2016) finds a positive correlation between disaster magnitude and crime rates across continental counties in the United States. Zahran et al. (2009) find that index, property, and violent crimes *decrease* while domestic crime *increases* in Florida after natural disasters. Breetzke et al. (2018) report that *overall* crime in Christchurch decreased after the Canterbury earthquake sequence (CES), but about 85 % of its neighborhoods experienced an increase in overall crimes. This paradoxical finding can be explained by closure of the Central Business District (CBD), which accounted for a large share (12 %) of *all* crime pre-CES and thus drove a large part of the reduction. After the CES, crime was *displaced* from usual hot spots to the rest of the city, as well as the neighborhoods. There was no pattern for neighborhoods that experienced increased crimes, but lower median income was found associated with one or more categories of crime increase (Breetzke et al. 2018), in Christchurch (unlike the other crimes). The actual number could be higher since domestic violence is often underreported.

The standard way to estimate social loss due to crime⁴⁷ is using the willingness-to-pay (WTP) approach.

Loss due to crime

 $= (WTP \text{ to reduce crime}) \times (Incidence \text{ of crime due to a disaster})$ (18)

The first component, *WTP*, has been extensively discussed in the literature. The *WTP* can be quantified by either revealed preference (e.g., Bishop and Murphy 2011; Pope and Pope 2012) or stated preference (e.g., Cohen 2004), as previously discussed. Cohen (2004) finds that investing \$100-\$150/year on crime control programs can reduce crime by 10 %. The Institute for Women's Policy Research (IWPR 2017) reports that the mean cost of medical care for those who sought treatment after a physical assault by an intimate partner was \$4,273 per incident in 2017 dollars. Of those seeking mental health services, there was an additional cost of \$1,631 per incident. Interested readers may refer to Doyle and Aizer (2018) for a comprehensive review of the state of the art in economics of domestic violence.

5.1.7. Indirect physical losses

Finally, there are other indirect costs that can be saved by adopting enhanced design, such as debris removal and management expenses, greenhouse gas emissions and energy use for building repairs, and environmental pollution associated with material production and waste disposal.

Debris removal. Debris after a disaster may include waste soils and sediments, vegetation (e.g., trees,

⁴⁶ In the context of a natural disaster, social disorganization theory would posit that crime increases in the ensuing chaos of response and recovery after an event. In contrast, routine activity theory would posit that increases in crime arise from disruptions to the routine activities of potential offenders, such as displacement or evacuation, which align to provide a motivated offender, a suitable target, and the absence of a capable guardian for the target (Zahran et al. 2014).

⁴⁷ The cost of crime to society is significant. OJP (1996) reported that victimizations generated \$105 billion annually in property and productivity losses and medical expenses in the United States, which means an annual "crime tax" of roughly \$425 per person. When adding in pain, long-term trauma, and risk of death, annual costs of crime could reach \$450 billion, or annual "crime tax" could be \$1,800 per person. A more recent study suggests that the annual cost of crime could be between \$690 billion and \$3.41 trillion in 2016 dollars (GAO 2017).

limbs, shrubs), municipal solid waste (e.g., common household garbage, personal belongings), construction and demolition debris (e.g., building materials, roads, bridges), vehicles (e.g., cars, trucks, boats), food waste, large home appliances (e.g., refrigerators, freezers, air conditioners), and household hazardous waste (e.g., cleaning agents, pesticides, pool chemicals). Each type of waste may contain or be contaminated with certain toxic or hazardous constituents (CRS 2011; EPA 2019). Common options to manage debris include landfilling, recycling, and burning. During the 1992 Hurricane Andrew, about 43 million cubic yards of debris were generated over a 500-square-mile area (CRS 2011). The 1994 Northridge Earthquake left about 7 million cubic yards of debris (CRS 2011). During the 2005 Hurricane Katrina, more than 99 million cubic yards of debris were generated, and debris removal alone cost more than \$3.7 billion (EPA 2019).

The costs for debris removal depend on the amount of debris and unit price for cleanup:

$$Debris removal costs = (Cleanup costs) \times (Amount of debris)$$
(19)

Following Hurricane Katrina, FEMA developed a Reasonable Cost Matrix for estimating debris removal costs and determining eligible reimbursement levels for contractors. The matrix provides unit prices for nine categories of debris in line with the FEMA 322 Public Assistance Guide (FEMA 2020). Note that debris removal is largely funded by FEMA and partially funded by state and local governments (CRS 2011; FEMA 2020).

Greenhouse gas emissions. The construction sector accounted for more than 11 % of annual global CO₂ emissions in 2018 (IEA 2019). Building repairs and demolition after a natural hazard event can result additional CO₂ emissions (Gonzalez et al. 2022). Comber et al. (2012) estimate that a 5-story concrete moment frame office building (75 000 sf/ 6 968 m², designed under current codes) could suffer damage resulting in an environmental impact of 2 820 tCO2e, or 19 % of the building's total embodied carbon footprint after a MCE (maximum considered earthquake) event in Seattle, WA. A 5-story concrete shear wall office building (75 000 sf/ 6 968 m², designed under current codes) could suffer damage resulting in an environmental impact of 2 260 tCO2e, or 15 % of the building's total embodied carbon footprint after a MCE event in Seattle, WA. In addition, a tilt-up concrete research laboratory (49 000 sf/ 4 552 m², constructed in 1963) could suffer damage resulting in an environmental impact of 7 320 tCO2e, or 31 % of the building's total embodied carbon footprint after a MCE event in Seattle, was an environmental impact of 7 320 tCO2e, or 31 % of the building's total embodied carbon footprint after a MCE event in San Francisco Bay Area.

The social cost of carbon (SCC) due to building damage and repairs depends on damage level and embodied carbon footprint of the building:

$$SCC = (Damage ratio) \times (Embodied carbon footprint) \times (Carbon price)$$
 (20)

Embodied carbon footprint refers to CO_2 emissions arising from the manufacturing, transportation, installation, maintenance, and disposal of building materials. In most cases, only emissions associated with materials are considered because data for other items are scarce (Comber et al. 2012).

Under the Executive Order 12866, the federal government's Interagency Working Group (IWG) developed an estimate for SCC, considering climate change impacts on agriculture, forestry, water, energy use, sea level rise, ecosystems, human health, and extreme weather (IWG 2016). The purpose of such estimate is to assist agencies to incorporate the social benefits of reducing CO₂ emissions into BCA

of regulatory actions. Table 5.2 presents the estimated SCC at different discount rates from 2010 to 2050.

Year of emission	Average estimate at 5 % discount rate	Average estimate at 3 % discount rate	Average estimate at 2.5 % discount rate	High impact estimate (95th percentile estimate at 3 % discount rate)
2010	10	31	50	86
2015	11	36	56	105
2020	12	42	62	23
2025	14	46	68	138
2030	16	50	73	152
2035	18	55	78	168
2040	21	60	84	183
2045	23	64	89	197
2050	26	69	95	212

Table 5.2. Social cost of carbon from 2010 to 2050, in 2007 dollars per metric ton of CO₂. Adapted from IWG (2016).

However, market-based carbon prices are much lower than the SCC estimates. So far, twelve U.S. states have enrolled into carbon pricing programs. The Regional Greenhouse Gas Initiative (RGGI) is a cooperative effort among eleven Eastern states to cap and reduce CO_2 emissions from the power sector. The minimum allowance price is \$2.38 per metric ton of CO_2 in 2021 (RGGI 2021). In California, the carbon pricing program involves power, building, transportation, and industry sectors. The minimum allowance price is \$17.71 per metric ton of CO_2 in 2021 (ICAP 2021).

Energy consumption. Building repairs may involve demolition and replacement of damaged components, temporal protection for affected areas, and relocation and reinstallation of equipment. Feese and Bulleit (2015) estimate that in Los Angeles, steel buildings built to high code standards (high code of Hazus) may sustain damage that incurs a total repair cost of \$38 226 and energy consumption of 33 565 MJ under a seismic event equivalent to the 1994 Northridge Earthquake. The annualized energy consumption for repairing a high-code steel building could be 1663 MJ. For a moderately designed (moderate code of Hazus) steel building, annualized energy consumption for repairs could be 4103 MJ. On the other hand, high-code concrete buildings may sustain damage that incurs a total repair cost of \$56 189 and energy consumption of 91 778 MJ under a seismic event equivalent to the Northridge

Earthquake. The annualized repair energy could be 606 MJ for a high-code concrete structure and 1968 MJ for a moderate-code concrete structure.

The energy cost due to building damage and repairs depends on damage level and embodied energy of the building:

 $Energy \ costs = (Damage \ ratio) \times (Embodied \ energy) \times (Energy \ price)$ (21)

Many databases are available for life-cycle assessment with respect to CO₂ emissions and energy consumption (e.g., Athena database, U.S. life cycle inventory database). A comprehensive review and comparison for those databases can be found at Martinez-Rocamora et al. (2016).

Environmental pollution. Material production and transportation, building construction, repair and demolition, as well as waste disposal (e.g., landfills, combustion), can discharge various pollutants into the environment. The most concerning environmental impacts include ozone depletion due to chlorofluorocarbons (CFC), photochemical oxidation due to ethylene (C_2H_4), eutrophication due to phosphorus concentration (PO₁⁻), and acidification due to sulfur dioxide (SO₂), as specified by the ISO 14025 Standard.

5.2. Estimating costs associated with enhanced building design

Table 5.3 provides a list of the potential costs associated with implementing recovery-based design, while accounting for the parties who bear the costs. The table also depicts common methods for cost estimation.

Table 5.3. Potential adoptio	n/implementation cost	s, who bears cost	t, and how to e	estimate co	st.

Cost	Who bears	Estimation methods	
New construction	Developer	 Construction cost data (e.g., RSMeans, Craftsman) Historical data Statistical models 	
Retrofit (hard and soft)	Owner (title holder)	 Construction cost data (e.g., RSMeans, Craftsman) Historical data Statistical models 	
Maintenance	Owner	 Life-cycle cost data (e.g., RSMeans) Historical data (e.g., ASHRAE Service Life and Maintenance Cost database) Statistical models 	
Code implementation	State and local governments	Historical data (e.g., code book purchase, education and training expenses, technical support, financial incentives)	
Plan evaluation	Owner	Covered by permit fees	
Site inspection	Owner	Covered by permit fees	
Construction permit application	Owner	% of construction or improvement costs	
Financing	Owner	(Interest rate) × Principal	
Additional cost to preserve historical or cultural characteristics	Government	Historical data (e.g., cost estimates, dedicated funds, financial incentives)	

Initial construction cost. The construction cost of a building comprises material, labor, and equipment costs, as well as contractor overhead and profits. Construction costs are influenced by the location of the project because the availability and costs of materials and labors, as well as regulatory requirements, vary from region to region. Construction costs also change over time due to inflation and other factors. While consulting firms, contractors, and agencies tend to develop their own codes for cost estimation, construction cost books, such as RSMeans and Craftsman, offer an alternative way to estimate costs based on a national database. Regional indices are provided by these books for quick adjustment from the national average to any cities.

Yet, there are two main constraints with use of construction cost books. First, cost data is collected on residential and commercial buildings nationwide, whereas public buildings (e.g., schools) and critical facilities (e.g., hospitals) are not included. In recovery-based design, the methods that critical facilities take to attain high performance objectives can be extended to designing residential and commercial buildings for higher performance. Lack of data for critical facilities may restrict the use of construction cost books for enhanced building design. Second, the data does not distinguish local difference in code requirements (e.g., design hazard levels, design loads), which can affect building design and construction costs. Understanding the impacts of local code requirements on construction costs is important to accurately estimate cost. Other methods for cost estimation are elaborated in Appendix B.

Structural improvement costs. The increased structural cost reflects increased material and labor costs for a building constructed with enhanced structural systems relative to the status quo. Structural costs are computed as unit prices of structural components multiplied by quantities. Improvement costs are estimated as the difference in structural costs before and after upgrades.

Structural costs =
$$\sum [(Unit price of a structural component) \times Quantity]$$
 (22)

Figure 2 illustrates the procedure to estimate structural costs using RSMeans data. Step 1 inventories the type and quantity of structural components in the building. Step 2 looks up the RSMeans assembly database for each type of structural component. If there is a matched assembly in the database with respect to materials, size, and maximum capacity, the total cost of this component type can be estimated by multiplying assembly cost and quantity. Otherwise step 3 is taken to assemble the costs for the component using the RSMeans unit cost database, or alternatively interpolate the costs from two similar assemblies. Once the unit cost is determined, the total cost is estimated by multiplying unit cost and quantity.



Figure 2. Structural cost estimating procedure.

Nonstructural improvement costs. There are three common approaches to estimating increased nonstructural costs.

Approach 1. Nonstructural improvement costs are estimated as a percentage of structural improvement costs.

Nonstructural improvement costs = $Percent \times (Structural improvement costs)$ (23)

The percentage is somewhere between 50 % and 100 %. Preston et al. (2019) report that on average, the cost of nonstructural upgrades (from NPC-2 to NPC-5) is 50 % of that of structural upgrades (from SPC-2 to SPC-4D) based on data from 45 hospitals in California. The ratio of nonstructural to structural costs ranges from 0.051 to 1.08. Fung et al. (2021b) suggests that nonstructural retrofit costs can be comparable to structural retrofit costs for residential buildings. Approach 1 is used when nonstructural improvements are not clearly defined but a coarse and handy estimate for nonstructural costs or total improvement cost is needed.

Approach 2. Nonstructural improvement costs are estimated as cost per square footage of upgrades times floor area.

Nonstructural improvement costs = (Cost per sqft of upgrades) \times (Floor area) (24)

This approach is used when nonstructural improvements are defined but lack design details. Thus, cost data for similar buildings (e.g., use category, location, improvements) are utilized to evaluate upgrade costs for the building.

Approach 3. Nonstructural costs are computed as unit prices of nonstructural components times quantities. Improvement costs are estimated as the difference in nonstructural costs before and after upgrades.

Nonstructural costs

$$= \sum [(Unit price of a nonstructural component) \times Quantity]$$
⁽²⁵⁾

Nonstructural components include stairs, elevators, cladding and glazing, distribution panels, ceilings, heating, ventilation, and air conditioning (HVAC) systems, which are important to the functions of a building. Notably, the Department of Health Care Access and Information (HCAI), formerly the Office of Statewide Health Planning and Development (OSHPD), has established a seismic certification program for nonstructural equipment (Special Seismic Certification Preapproval, OSP) and anchorages (Preapproval of Anchorage, OPA) used in health facilities in California. The pre-approved equipment and anchorages can bear higher seismic loads and ensure continuing operation of hospitals after a major earthquake. Detailed reports for those components are provided on HCAI's website, as well as contact information for manufactures, providers, and suppliers (HCAI 2022). However, the cost premiums for those components are not available in construction cost books. Moreover, emergency systems may be revamped to achieve functional recovery goals. This may involve emergency equipment, alarms, messaging, and emergency communication plans or strategies in support of organizational resilience (Almfuti and Wilford 2013).

Approach 3 can provide the most accurate cost estimates, but it also takes the greatest effort to collect cost data, including but not limited to conversations with engineering professionals, consultants, and manufacturers, as some of those data are not publicly accessible. Moreover, detailed information for nonstructural design is needed when this approach is employed.

Life-cycle cost. The life-cycle cost (LCC) is the total cost associated with building design and construction, building operation and maintenance, and building disposal at the end of the life cycle. FEMA's BCA guide recommends considering at least a replacement for nonstructural components in the middle of the life cycle when both nonstructural and structural improvements are analyzed (FEMA 2009). This is reasonable because the useful life of nonstructural components is between 15 and 30 years, less than that of structural components. The cost for replacing nonstructural components is an important portion of maintenance costs. Yet, maintenance costs also include the costs for scheduled maintenance to keep mechanical systems in working order, preventative maintenance to reduce the likelihood of system failure, and actual repairs of building's mechanical, electrical, and plumbing systems. Overall, maintenance costs are flexible and dependent on many factors other than the building itself (e.g., budget).

There might be questions on how recovery-based design affects maintenance costs over the life of the building. For example, are nonstructural components more robust to maintenance, reducing maintenance costs? Does building in redundancy add to maintenance costs? To answer these questions, more data is needed to determine whether the divergence in costs is large enough to affect decision making. Moreover, such maintenance cost impacts are not benefits in and of themselves, as discussed in Sec. 3, should be subtracted from benefits.

5.3. Estimating building downtime

A key step of the FR-BCA process is quantifying the benefits of functional recovery design in terms of avoided losses for mitigated and unmitigated alternatives, as discussed above. To do so requires an analytical method capable of quantifying expected building performance into important direct and indirect losses such as building repair costs and downtimes. Additionally, the method should be detailed enough to capture differences in performance between selected mitigation alternatives.

This section reviews available methods and software that can be used to estimate building repair costs and downtime and key advantages and disadvantages of each approach, as well as gaps in the state-of-the-art. Four of the most commonly used methods are discussed in the preceding sections and summarized below in Table 5.4. Among the four methods discussed, the specific definition of building functional recovery can vary. While additional recovery-based performance states and definitions can be found in other literature (e.g., Comerio 2006; Burton et al. 2015; Mieler and Mitrani-Reiser 2018), most of the methods discussed here quantify recovery time in terms of re-occupancy, functional recovery, and full recovery, where re-occupancy means the building is safe to enter and able to provide shelter and functional recovery means the building may not be fully repaired but has enough capacity to serve its basic intended function. The downtime assessment methods presented here quantify losses and repair times at the building level, focusing on the performance of the building itself, rather than influences from surrounding infrastructure, regional recovery policy, or individual household or business capacities to function outside the building.

Method	FEMA P-58	REDi	Hazus	ATC 138*
Level of Modeling	Component	Component	Building	Component and Building Systems
Scope of recovery assessment	Recovery time due to building damage alone	Recovery time includes time to repair building damage, post-earthquake inspection, engineering mobilization, financing, contractor mobilization, permitting, and utility downtime.	Recovery time includes time for decision making, building construction and clean-up, and time to obtain financing, permits and complete design.	Recovery time includes time to repair building damage, post- earthquake inspection, engineering mobilization, engineering design, financing, contractor mobilization, and permitting.
Repair time model	Building repair time is estimated from simulated component repair times using simplified serial and parallel worker allocation assumptions.	Building repair time is estimated from simulated component repair times using a repair scheduling algorithm which considers workers allocations limits and repair sequence constraints	Based on heuristic and empirical estimates of building repair time.	Building repair time is estimated from simulated component repair times using a repair scheduling algorithm which considers workers allocations limits and repair sequence constraints
Building Function Assessment	No explicit assessment of building function.	Building regains function when the repair schedule for all RC2 and RC3 components is complete.	Judgment-based multipliers to determine whether building repair interrupts business operation (e.g., relocation).	Each tenant unit within the building regains function when the building is safe, accessible, and the required systems become acceptably operational for basic function.

Table 5.4. Assessment methods for downtime/functional recovery time estimation.

Basis of recovery and repair time data	<i>Fragility database</i> provides empirical recovery time for more than 700 structural and nonstructural components at various damage states.	<i>Fragility database</i> provides empirical recovery time for more than 700 structural and nonstructural components at various damage states.	<i>Consequence tables</i> provide median recovery time estimates for 33 occupancy classes at five building damage states (none, slight, moderate, extensive, complete). A set of <i>multipliers</i> are used to adjust recovery time to account for different building size and financing difficulty.	<i>Fragility database</i> provides empirical recovery time for more than 700 structural and nonstructural components at various damage states.
Key Limitations	Does not explicitly consider building function.	Recovery time tends to be overestimated because too many impediments are considered in the recovery model (Terzic and Kolozvari 2020). Building function model is not occupancy specific and does not consider the operations of building systems.	Only median recovery time is used, without upper and lower bounds that account for uncertainties. Outcomes limited to the building class level and are not capable of capturing certain building- specific response characteristics.	When building-specific data is not available, assumptions regarding tenant requirements for basic intended function and system operations are based on engineering judgment.
Key Advantages	Building specific and probabilistically robust.	Incorporates consideration of impeding factors and more advanced repair scheduling.	Computationally inexpensive.	Explicitly models system operation to quantify building function

Note: RC = Recovery class. See Sec. 5.2.3 for details.

*ATC-138 is an ongoing project led by FEMA and Applied Technical Council (ATC), which will establish recovery-based objectives for highperformance buildings. There are several publicly available and proprietary software packages that facilitate the estimation of functional recovery times for buildings and building portfolios according to these four methodologies as shown in Table 5.5. For additional review of the state-of-the-art in this rapidly evolving field, see Meiler and Mitrani-Reiser (2018) and Cook et al. (2022).

T 11 C C	0 0	C	1	10 1		· •	,• ,•
lable	Soffware ni	roorams for	downfime	(functional	recovery	time e	stimation
1 4010 5.5.	Solution pr	ograms for		ranetional	1000,019		Stilliation

Software	PACT	SP3	Hazus	PELICUN
Assessment Method(s)	FEMA P-58	FEMA P-58 REDi ATC-138	Hazus	FEMA P-58
Platform	Publicly available local software package	Proprietary web- based analysis platform	Publicly available local GIS-based software package	Open-source python software package hosted on web-based analysis platform
Publisher	FEMA	Haselton Baker Risk Group	FEMA	SimCenter

5.3.1. Hazus

To date, the most common method of assessing building recovery time in the literature, especially when evaluating the recovery of a community or region, has been through the use of recovery functions such as those from Hazus (FEMA 2012). Hazus was developed as part of the FEMA Multi-hazard Loss Estimation Methodology for estimating seismic risk to U.S. infrastructure. Hazus includes prediction models for direct physical damage and loss, induced physical damage, direct economic and social losses, and indirect economic losses.

To develop loss predictions, the Hazus methodology combines site hazards with predefined fragility and consequence functions for a given building class. Building classifications in Hazus consist of four building attributes: building type (e.g., wood light frame or concrete moment frame), height classification (low-rise, mid-rise, and high-rise), level of code (based on the building's location and age), and occupancy (e.g., commercial office or single-family dwelling). Hazus uses the capacity spectrum method to estimate the building's spectral accelerations and spectral displacements for a given level of shaking, which are in turn used to quantify the building's damage state (none, slight, moderate, extensive, or complete) and damage consequences (e.g., repair cost or recovery time) from the predefined fragility and consequence functions, respectively. While building-specific inputs can be used to adjust the building's capacity spectrum, such as building period and base shear strength, typical values are provided for each building class to facilitate the estimation of building response and damage.

To estimate building recovery and service interruption time, Hazus defines building-level repair and recovery times as a function of occupancy class and damage state. Note that building recovery times consider construction and clean up time as well as time to obtain financing, permits and complete design, whereas building repair times only consider construction and clean up time. In addition, Hazus uses a set of multipliers to adjust recovery time to account for different building size and financing difficulty.

The fragility and consequence functions developed in Hazus are based on a mix of historical data, previous research, and engineering judgment. In particular, the method heavily relies on the MMI damage and loss functions and building classifications developed as part of the ATC-13 project (ATC 1985), which were primarily based on expert judgment. While the repair costs functions were calibrated to empirical data from the Long Beach (1993), San Fernando (1971), Coalinga (1983), Morgan Hill (1984), Loma Prieta (1989), and Northridge (1994) earthquakes (Kircher 2006), all repair and recovery time and loss of function modifiers provided in Hazus are based solely on engineering judgement.

While empirical and judgment-based building-class-level recovery functions are useful to aggregate risk information from large building stocks, they typically are unable to capture building-specific attributes of performance necessary to compare the performance of two buildings with unique design characteristics. For example, since Hazus classifies buildings into one of three building age categories, pre-1941, 1941 to 1973, and post-1973, any building designed after 1973, including those designed after the Hazus release of 1997, are estimated to have the same performance as any post-1973 buildings of the same class, regardless of building design features.

5.3.2. FEMA P-58

Performance-based earthquake engineering (PBEE) is a statistically robust means of quantifying and communicating building performance into resilience metrics, such as repair costs or functional recovery. The approach integrates a probabilistic hazard analysis, with a structural response assessment and a component-level damage assessment to quantify the damage to each structural and nonstructural component within the building for a given shaking intensity (Porter 2003; Moehle and Deierlein 2004). The performance outcomes from a performance-based assessment depend on the building's specific design and configuration characteristics and can be used to facilitate the comparison of design alternatives and mitigation strategies for new resilience-based performance objectives.

FEMA P-58 (FEMA 2018) is one of the most commonly used implementations of the PBEE methodology and probabilistically quantifies building performance based on its structural and nonstructural characteristics in terms of repair costs, repair time, casualties, and unsafe placards. To facilitate the component-level damage simulations, FEMA P-58 collects a database of over 700 component fragility and consequence functions for various structural and nonstructural building components based on empirical earthquake data, experimental testing, analytical analysis, and engineering judgment. Building repair times are quantified as an aggregation of the estimated time a worker will take to repair damage to each component within the building; component-level repair times are aggregated based on a repair schedule that assumes all workers will either repair all stories at once (parallel) or one story at a time (series).

Compared to the building-class recovery times estimated by Hazus, FEMA P-58 provides detailed repair time estimates based on building-specific response and component characteristics, but requires significantly more detailed assessment inputs to define the building-specific characteristics and is computationally more

expensive to run (depending on the structural analysis method used to define building response). In its current form, FEMA P-58 only quantifies building repair times due to direct building damage and does not explicitly consider the recovery of building function (which differs from the repair time of all components in the building) or external factors that may impede the start of construction, however, previous studies have used the story-based repair time estimates from FEMA P-58 as an upper bound of building functional recovery time (Cimellaro and Piqué 2016). Building repair times from FEMA P-58 are also based on a simplified repair schedule algorithm, which does not consider critical constraints when estimating building-level repair times.

5.3.3. REDi and REDi-based methods

The Resilience-based Earthquake Design Initiative for the Next Generation of Buildings (REDi), expands on the FEMA P-58 methodology to quantify, probabilistically, building re-occupancy and functional recovery times (Almufti and Willford 2013). Additionally, the REDi methodology improves on some of the key limitations of the FEMA P-58 framework by incorporating estimates of pre-repair impeding factors, such as time time required to gather finances or obtain building permits; the REDi method replaces the simplified FEMA P-58 repair schedule algorithm with a more sophisticated repair sequence, which borrows key concepts form the critical path method (Kelley and Walker 1959) to schedule repairs of damaged components based on worker limitations and component repair sequence constraints.

To facilitate the quantification of building re-occupancy and functional recovery times, REDi assigns a Repair Class (RC) to each damage state of every component within the FEMA P-58 fragility database. The Repair Classes define the impact the component's damage has on the building performance, where RC1 represents minor or cosmetic damage that hinders full recovery; RC2 represents more significant damage which hinders building function; and RC3 represents heavily damaged components that pose a life-safety risk and hinder building reoccupancy. The total building functional recovery time is estimated using a repair schedule to repair damage to all components with RC2 or RC3 level damage.

The REDi method has been used in the performance-based design of new buildings, such as the 181 Fremont Tower in San Francisco (Almufti et al. 2016), and as part of the U.S. Resiliency Council's seismic rating system (USRC 2015). However, the REDi method oversimplifies the quantification of functional recovery; instead of explicitly modeling the effect component damage on the operation of building systems, REDi bases the building's functional recovery time on the performance of the "worst-case-component" (e.g., damage to a single component flagged as affecting function will cause a loss of function of the entire building). Additionally, previous studies have noted that REDi tends to provide overly conservative estimates of recovery time for cases with low levels of damage, due to a low threshold for triggering high consequence impeding factors and other impediments considered in the recovery model (Paul et al. 2018; Terzic and Kolozvari 2020).

Several studies have proposed modified frameworks to address key limitations for assessing functional recovery in the REDi methodology. Notably, Paul et al. (2018) proposes updates to the repair class and impeding factor logic to improve the sensitivity to triggering large consequences for low damage states and a more holistic consideration of assessment uncertainties. Additionally, Molina-Hutt et al. (2022) proposes a new framework built upon the REDi methodology. The framework defines two new building performance states for building stability and building shelter, incorporates functions to estimate delays due to temporary

repairs and demolition time, and includes a new repair schedule algorithm to estimate the recovery of function of each story in the building.

5.3.4. ATC-138 and fault tree-based methods

Key to quantifying a building's functional performance is the capability of an analytical method to model the operations of a building's various structural and nonstructural systems and their subsequent effect on tenant function. Porter and Ramer (2012) introduce the idea of applying fault trees (Fussell et al. 1974) to model the reliability of a building system to assess post-earthquake downtime of a California data center, representing a robust framework to model the various dependencies among building systems and explicitly quantify overall building function. Recently, Cook et al. (2022) and Terzic and Villanueva (2021) propose new assessment methodologies for the quantification of building functional recovery, which introduce fault tree models to explicitly model the operation of various structural and nonstructural buildings within the FEMA P-58 assessment framework. Both studies utilize the FEMA P-58 fragility and consequences database to facilitate the simulation of building damage, and contribute to improved repair scheduling algorithms to estimate recovery times.

The ATC 138 project is an ongoing FEMA-funded project which adopts the functional recovery assessment framework proposed by Cook et al. (2022), as part of a NIST-Funded effort, into the FEMA P-58 methodology to establish recovery-based design tools and guidelines. The adopted methodology provides the various default system-level fault trees, component-level attributes, and tenant-specific requirements to model the re-occupancy and functional recovery times for modern U.S. buildings. In most cases, these default attributes are based on engineering judgment, given the lack of empirical and experimental data. The repair schedule algorithm includes the consideration of temporary repairs that mitigate near-term impact of component and system failures on building function and quantifies external factors that delay the start of repair such as inspection, permitting, gathering financing, and engineering and contractor mobilization (similar to REDi). A computational implementation of the method is currently accessible from the proprietary software SP3 (HB-Risk 2022) and a MATLAB source code (Cook 2022).

5.4. Case study

To illustrate the application of FR-BCA, we present a case study example of recovery-based design options for new construction. The case study is not intended to be exhaustive or suggestive of a design recommendation for practice. More details can be found in Fung et al. (2022) and upcoming follow-up publications.

Step 0. Since the case study is illustrative, we consider a conservative discount rate, $\delta = 0.07$, and planning horizon, T = 50 years, as well as a more lenient discount rate, $\delta = 0.02$, and planning horizon, T = 75 years. In addition, our baseline design is an archetype 4-story reinforced concrete moment frame (RCMF) commercial office building designed per ASCE 7-16 (B-4), for a generic site with Seismic Design Category D (S_{ds} = 1.0g and S_{d1} = 0.6g) and Site Class C (Cook and Sattar 2022). The three recovery-based design options include:

- structural improvements only (S);
- nonstructural improvements only (NS); and
- a design that incorporates both structural and nonstructural improvements (S-NS).

Structural improvements include redesigning the sizing and reinforcement of the components of the lateral system to increase design strength and reduce lateral drifts. Nonstructural improvements include modifications of the building model to include the effects of increased anchorage and bracing, the use of seismically rated equipment, and several other design modifications to reduce the vulnerability of certain nonstructural systems. It should be noted that these designs are largely limited to increasing capacity and do not consider other strategies (e.g., base isolation, resilience planning).

Step 1. We consider a subset of the avoided losses presented in Table 5.1. Specifically, we consider repair costs, repair time, time to reoccupancy, and time to functional recovery (all results of performance assessment), as well as costs associated with occupant displacement and loss of business and rental incomes (due to business interruption). Direct losses are calculated using the probabilistic performance-based assessment method described in Cook and Sattar (2022). Indirect losses due to occupant displacement include the costs to provide temporary accommodation and the value of lost income (IMDC 2020); business income is estimated from proprietor income (BEA 2020); rental income is estimated from the national average rental income for office buildings (BOMA 2018).

Given the research needs required for some of the other categories of losses in Table 5.1, we include those that are the least controversial. While the inclusion of a broader set of avoided losses would certainly improve the analysis, determining reasonable assumptions for their estimation is beyond the scope of this case study example.

Step 2. Finally, we estimate investment costs for the four archetype designs. Investment costs are based on RSMeans (2021) as well as conversations with design professionals. It should be noted that these costs are not intended to reflect construction costs in practice. However, based on conversations with design professionals, as well as trends in the construction industry, the cost *differences* between the baseline and recovery-based designs are plausible (that is, within bounds of average expected costs). For a BCA, this difference in investment costs represents the cost of a recovery-based design option.

Step 3. Given avoided losses and investment costs, we compute PV(benefits), the present value of expected annual net benefits, and PV(costs), the present value of additional investment costs relative to the baseline design. Note that while the definition of PV(costs) is specific to new construction, the formula for BCR is generic (for existing buildings, PV(costs) would be the retrofit costs as the baseline option is to do nothing). We then compute the benefit-cost ratio (BCR), defined as BCR = PV(benefits) / PV(costs), and the net present value (NPV), defined as NPV = PV(benefits) - PV(costs). Table 5.6 presents the results of the benefit-cost analysis for the three candidate recovery-based designs.

Table 5.6. Case study benefit-cos	t analysis results (relative to	o baseline design). Low: a	$\delta = 0.07, T = 50.$
High: $\delta = 0.02, T = 75.$			

Design	BCR (Low)	BCR (High)	NPV (Low)	NPV (High)
S	0.093	0.261	-\$571 096	-\$465 492
NS	1.412	4.071	\$56 684	\$410 584
S-NS	0.380	1.069	-\$475 948	\$52 888

The case study suggests that nonstructural improvements alone may provide cost-effective design options for achieving recovery-based performance objectives. The relatively large BCRs for NS (ranges from 1.412 to 4.045) are due to the relatively low costs of implementing nonstructural improvements compared to the reductions in avoided loss associated with the improved robustness of the nonstructural systems. Structural improvements alone, on the other hand, do not appear to be cost effective for recovery-based objectives due to the substantial structural construction cost increases relative to marginal avoided losses at frequent hazard levels. Using the decision criterion BCR > 1 (i.e., the benefits outweigh the costs) or NPV > 0 (i.e., the value of the investment exceeds its cost), we are most likely to choose NS out of the three case-study designs, which saves between \$1.41 and \$ 4.07 per dollar invested. Note, however, that benefit-cost analysis outcomes and recovery-based design solutions may vary significantly from this case study for different levels of seismicity and varied building characteristics. In forthcoming companion publications, we provide the details of benefit and cost estimation for several archetypes and structural systems.

Step 4. We now apply the benefit-transfer matrix presented in Table 4.1 to distribute benefits and costs. Given the results of Step 3, we focus on the NS archetype. Note that we only consider the first four columns of Table 4.1, which represent construction cost, property loss, and direct business interruption, and indirect business interruption. In addition, we distribute rental income losses to the title holder and displacement costs to the tenant. The remaining columns (i.e., insurance and deaths/injuries) are not covered by our case study. The results are shown in Table 5.7.

Stakeholder	Weights (benefits)	Weights (costs)	Distributed benefits	Distributed costs	Distributed BCR	Distributed NPV
Developer	0.02 / 5		776.86		-	776.86
Title holder	(0.58 + 1) / 5	0.5	61 372.14	68 765.77	0.89	-7393.63
Lender	0.07 / 5		2719.02		-	2719.02
Tenant	(0.33 + 1 + 1) / 5	0.5	90 504.49	68 765.77	1.32	21 738.72
Community	1 / 5		38 843.13		-	38 843.13

Table 5.7. Distributed benefit-cost analysis for archetype NS, based on the benefit-transfer matrix in Table 4.1 (NIBS 2019). Weights for benefits are unnormalized.

Since we apply the benefit-transfer matrix *ex post*, we sum the weights for benefits across columns and normalize so that weights sum to one. Note that we apply the table as given, assuming that the title holder bears the cost for new construction and passes 50 % of that cost to the tenant. It might be more reasonable to distribute costs between the developer and title holder instead. Moreover, based on the review of the literature on indirect economic losses due to supply chain disruptions, we assume a conservative multiplier factor of 5 to scale direct business interruption losses to the community.⁴⁸

Note that the BCR is not defined for developers, lenders, and the community since their cost is zero. On the other hand, the title holder and tenant face a BCR of 0.89 and 1.32, respectively. This example illustrates the concern that recovery-based design disproportionately benefits the stakeholders that bear less (or none) of the cost. The NPV, which is well-defined for all stakeholders, emphasizes this point more clearly.

⁴⁸ While there is not a lot of data on these factors, Inoue and Todo (2019) suggest a multiplier factor of about 20.

6. Conclusion

6.1. Summary/practical guidance

Economic evaluation is a key step toward developing standards and policies. Building codes define the minimum requirements for new construction, with the objective to safeguard the health and safety of occupants through affordable means. Retrofit policies target the most vulnerable portions of the building, in order to improve building performance while limiting retrofit costs. For recovery-based design, economic evaluation is especially important since no design criteria currently exist and functional recovery may have large-scale impacts, from building owners and occupants to the region and nation through supply chains. For individual decision makers, economic evaluation is a useful tool for comparing alternative strategies and assessing the feasibility of an investment.

This report provides a roadmap for how to evaluate benefits and costs for recovery-based design. The roadmap consists of four steps: set analysis parameters – estimate benefits – estimate costs – allocate benefits and costs to stakeholders. In each step, we provide a review of the methods and tools for performing the analysis.

- Estimate benefits. Twenty-seven benefit categories relevant to recovery-based design are described in Sec 5.1. Methods and data sources that can be utilized to quantify these benefits are elaborated with one or more examples, while highlighting opportunities for further research.
- Estimate costs. The approaches to estimating initial construction costs, structural improvement costs, nonstructural improvement costs, and life-cycle costs are reviewed in Sec 5.1. Appendix B provides additional methods for cost estimation.
- Allocate benefits and costs to stakeholders. The potential beneficiaries of each benefit category are enumerated in Table 5.1. Section 4 explains how to distribute benefits and costs to various stakeholder groups using a benefit transfer matrix (NIBS 2019).

While this roadmap is specific to recovery-based design, the principles may apply more broadly to other new design paradigms.

This report also elaborates the advantages and limitations of five economic evaluation methods in Section 3: benefit-cost analysis, cost-effectiveness analysis, life-cycle cost analysis, willingness-to-pay (WTP) methods, and economic impact analysis. Readers may refer to this guidance to determine the proper method to use.

6.2. Major findings

Based on an extensive literature review, we find that many indirect benefit categories are not well investigated in previous studies, even though these indirect benefits are highly relevant to community resilience, social equity, and environmental sustainability. In addition, the data required to quantify these benefits is scarce even though the methods (e.g., WTP) are readily available.

• Among the 27 benefit categories, three direct benefits are extensively assessed in the literature and considered as primary metrics for BCA: building damage, content damage, and casualties.

- Six indirect benefits are well studied and frequently included in BCA: avoiding displacement and debris removal, avoiding loss of business or rental incomes, avoiding loss of life quality, preserving culture/history/character, and avoiding energy consumption and greenhouse gasses emissions incurred by building repairs or demolition.
- Five benefit categories are investigated less within the literature but may have great influence on property owners and business owners: avoided loss of productivity, avoided reduction in customers, decrease of property value, increase in insurance premium, and supply chain delays.
- Seven benefit categories are rarely studied but may have profound influence on a broader community: avoided reduction in employment, tax base, and affordable housing, improved equity through reduction in regressive hazard insurance, reduced crime rate, and avoided government assistance in building repairs, inspection, and displacement of residents.
- Five benefit categories are rarely assessed in the literature due to high complexity and uncertainty: avoided delays in recouping return on investment due to delays, delays in settling insurance claims, litigation costs,⁴⁹ loss of sense of connectedness or community, and reduction in perceived risk due to building's high performance.

The case study for reinforced concrete moment frame (RCMF) buildings is part of our ongoing effort to measure the cost-effectiveness of recovery-based design. Other building types, such as buckling-restrained braced frame (BRBF) and steel moment frame (SMF), will be analyzed subsequently to provide a more holistic perspective of cost effectiveness. The preliminary results suggest that nonstructural improvements can contribute to a large proportion of annual avoided loss and can be more cost-effective than structural improvements alone. This implies a promising way to achieve functional recovery goals through nonstructural enhancement.

The case study also contributes to improved methods for benefit-cost analysis by taking nonstructural improvements and building-specific attributes into account. Previous studies often focus on enhancing structural systems while ignoring or underestimating the role of nonstructural enhancement. In less intensive earthquake events, nonstructural damage may cost more than that of structural damage when buildings are designed under the current code. This study takes the first step to compare the cost-effectiveness of three design strategies: structural improvements only, nonstructural improvements only, and combined structural and nonstructural improvements. As noted before, enhancing nonstructural components could improve building overall performance and produce greater net benefits compared to structural enhancement.

Furthermore, the case study utilizes a state-of-the-art performance-based assessment method to predict downtime for commercial office buildings (Cook et al. 2022). Building characteristics, occupancy categories, and performance objectives are addressed in the modeling stage and reflected in benefit estimates. The improved estimates could enhance decision making for a specific building given that

⁴⁹ Assuming a building is code-conforming, the owner/title holder has limited liability in case of damaging event. However, that does not preclude litigation, hence it is difficult to predict (avoided) litigation costs; moreover, litigation could result in additional delays in resuming operations, which would result in additional business interruption losses. Given the speculative and understudied nature of this category, we do not cover it in Table 5.1.

conventional estimation methods rely heavily on empirical and expert judgements, which are helpful to study a large building stock but are not accurate enough for a single building.

6.3. Future directions

Benefit-cost analysis framework. We provide a first step toward enumerating the range of potential benefits, as well as stakeholders and beneficiaries, associated with building functional recovery; however, this framework is not comprehensive. Future studies could explore co-benefits (i.e., benefits accrued even if no event occurs) and externalities (e.g., damage to other buildings) that may influence decision making for enhanced building design. Such co-benefits are understudied and present an opportunity for future research to enhance the business case for functional recovery. Another limitation of the framework is that the enumerated benefits may overlap or strongly correlate with other types of benefits. Thus, an ontology of benefits (classes of benefits, stakeholders, quantification) could be useful to unify this framework across various domains, which could be investigated in future work.

Case study for recovery-based design. We assess five benefit categories in the case study: reduced structural and nonstructural repair costs, avoided displacement costs, and avoided business and rental interruption costs. More benefit items that involve a broader range of stakeholders could be included in future analyses. In addition, there is a need for validated methods for distributing benefits and costs that build on Table 4.1 (NIBS 2019). Future studies may also aggregate results from the asset level to community level to provide a holistic view of the cost-effectiveness of alternative design strategies. In addition, we analyze three design strategies without controlling performance levels (e.g., repair time), improvement costs, or total benefits. While optimizing design strategies is not a priority at this stage, future studies could control one or two of the three variables and optimize the rest of variables, such as minimizing improvement costs for a target performance level or maximizing total benefits for a fixed improvement budget. Finally, beyond *designing* for functional recovery, there will be practical challenges with implementing functional recovery (including review and quality control) that are associated with non-negligible costs (Tokas 2011). Thus, practical recovery-based design will require a more holistic approach.

Existing buildings, critical infrastructure, and community impacts. While our case study is focused on new construction, future work could examine existing buildings and critical infrastructure. Economic evaluations of these systems may produce different results because existing building conditions could limit the capacity of improvements, and critical infrastructure may have greater impacts on the community. Moreover, the recovery of building functions depends on external conditions such as lifelines, which adds complexity to downtime estimation. Finally, it is important to understand and evaluate the impacts of recovery-based design standards for underserved communities. Improved building standards could enhance community resilience but at the same time increase housing and rental prices. Underserved communities are more likely to be affected because owners often pass increased construction costs to tenants through rents (ATC 2009).

References

- Abrahams et al. (2021) "NIST SP 1269: NIST-FEMA Post-Earthquake Functional Recovery Workshop Report" <u>https://nvlpubs.nist.gov/nistpubs/SpecialPublications/NIST.SP.1269.pdf</u>
- ACEHR (Advisory Committee on Earthquake Hazards Reduction). (2019) Effectiveness of the National Earthquake Hazards Reduction Program: A Report from the Advisory Committee on Earthquake Hazards Reduction. <u>https://www.nehrp.gov/pdf/September_2019_ReporttotheNISTDirector.pdf</u>
- ACI (American Concrete Institute). (2021) ACI 318 Building Code Portal. https://www.concrete.org/tools/318buildingcodeportal.aspx.aspx
- Acosta, R.J., Kishore, N., Irizarry, R.A., and Buckee, C.O. (2020) Quantifying the dynamics of migration after Hurricane Maria in Puerto Rico. PNAS, 117(51), 32772-32778.
- Ahlfeldt, G. M., and Maennig, W. (2010). Substitutability and Complementarity of Urban Amenities: External Effects of Built Heritage in Berlin. *Real Estate Economics*, 38(2), 285–323. <u>https://doi.org/10.1111/j.1540-6229.2010.00268.x</u>.
- AISC (American Institute of Steel Construction). (2021) Specification for Structural Steel Buildings. <u>https://www.aisc.org/Specification-for-Structural-Steel-Buildings-ANSIAISC-360-16-1</u>
- Alberini, A., and Longo, A. (2006). Combining the travel cost and contingent behavior methods to value cultural heritage sites: Evidence from Armenia. *Journal of Cultural Economics*, 30(4), 287–304. <u>https://doi.org/10.1007/s10824-006-9020-9</u>.
- Almufti, I., and Wilford, M. (2013) REDi Rating System: Resilience-based Earthquake Design Initiative for the Next Generation of Buildings. ARUP.
- Almufti, I., Krolicki, J., and Crowther, A. (2016) The Resilient-Based Design of the 181 Fremont Tower *Stucture Magazine*. <u>https://www.structuremag.org/wp-content/uploads/2016/05/F-</u> <u>FremontTowers-Jun16.pdf</u>
- Andersson, M., Kopsch, F., and Palm, P. (2019). How cultural values are reflected on the housing market direct effects and the cultural spillover. *International Journal of Housing Markets and Analysis*, 12(3), 405–423. <u>https://doi.org/10.1108/IJHMA-02-2018-0016</u>.
- Arrow, K. J., Cropper, M. L., Eads, G. C., Hahn, R. W., Lave, L. B., Noll, R. G., ... & Stavins, R. N. (1996). Is there a role for benefit-cost analysis in environmental, health, and safety regulation? *Science*, 272(5259), 221-222.
- ASCE (American Society of Civil Engineers). (2017a) Minimum Design Loads and Associated Criteria for Buildings and Other Structures (7-16) https://sp360.asce.org/PersonifyEbusiness/Merchandise/Product-Details/productId/233133882
- ASCE (American Society of Civil Engineers). (2017b) Seismic Evaluation and Retrofit of Existing Buildings (41-17) <u>https://sp360.asce.org/PersonifyEbusiness/Merchandise/Product-</u> Details/productId/233163464? ga=2.173930147.1078354897.1643664512-<u>1726119433.1628000448</u>
- ASCE (American Society of Civil Engineers). (2021) Hazard-Resilient Infrastructure: Analysis and Design. Infrastructure Resilience Division of ASCE. Reston, VA: American Society of Civil Engineers. <u>https://doi.org/10.1061/9780784415757</u>.
- ASTM International. (2020a). E1057-15(2020)e1 Standard Practice for Measuring Internal Rate of Return and Adjusted Internal Rate of Return for Investments in Buildings and Building Systems. Retrieved from <u>https://doi.org/10.1520/E1057-15R20E01</u>.
- ASTM International. (2020b). E1074-15(2020)e1 Standard Practice for Measuring Net Benefits and Net Savings for Investments in Buildings and Building Systems. Retrieved from https://doi.org/10.1520/E1074-15R20E01.
- ASTM International. (2020c). E1121-15(2020)e1 Standard Practice for Measuring Payback for Investments in Buildings and Building Systems. Retrieved from <u>https://doi.org/10.1520/E1121-15R20E01</u>.

- ASTM International. (2020d). E1185-15(2020)e1 Standard Guide for Selecting Economic Methods for Evaluating Investments in Buildings and Building Systems. Retrieved from https://doi.org/10.1520/E1185-15R20E01.
- ASTM International. (2020e). E2506-15(2020)e1Standard Guide for Developing a Cost-Effective Risk Mitigation Plan for New and Existing Constructed Facilities. Retrieved from <u>https://doi.org/10.1520/E2506-15R20E01</u>.
- ASTM International. (2020f). E964-15(2020)e1 Standard Practice for Measuring Benefit-to-Cost and Savings-to-Investment Ratios for Buildings and Building Systems. Retrieved from <u>https://doi.org/10.1520/E0964-15R20E01</u>.
- ATC (Applied Technology Council). (1985) Earthquake Damage Evaluation Data for California (ATC-13). <u>https://www.atcouncil.org/pdfs/atc13.pdf</u>
- ATC (Applied Technology Council). (2009) Here Today-Here Tomorrow: The Road to Earthquake Resilience in San Francisco, Earthquake Safety for Soft-Story Buildings. Report ATC 52-3, Applied Technology Council, Redwood City, CA. https://sfgov.org/sfc/sites/default/files/ESIP/FileCenter/Documents/9756-atc523.pdf
- Bade, D., Castillo, J. G., Fernandez, M. A., and Aguilar-Bohorquez, J. (2020). The price premium of heritage in the housing market: Evidence from Auckland, New Zealand. *Land Use Policy*, 99, 105042. <u>https://doi.org/10.1016/j.landusepol.2020.105042</u>.
- Bar-Gill, O. (2020). Willingness-to-Pay: A Welfarist Reassessment. SSRN Scholarly Paper No. ID 3700351. Social Science Research Network, Rochester, NY. <u>https://doi.org/10.2139/ssrn.3700351</u>.
- Bayer, K (2013) Census 2013: Christchurch's population slumps. <u>https://www.nzherald.co.nz/nz/census-</u>2013-christchurchs-population-slumps/GSC6C4YV6SQA3Q6AS2232BW4GQ/
- BEA (2018a). Bureau of Economic Analysis Glossary: Proprietors' Income. https://www.bea.gov/help/glossary/proprietors-income
- BEA (2018b). Bureau of Economic Analysis Glossary: Value Added. <u>https://www.bea.gov/help/glossary/value-added</u>
- BEA (2020). CAINC5N Personal Income by Major Component and Earnings by NAICS Industry.
- Beck, J., Kiremidjian, A., Mader, G., and Reitherman, R. (1997). Year One Research Report: Social, Economic, And System Aspects of Earthquake Recovery and Reconstruction. https://curee.org/projects/CK/phase-III/reports/CKIII-01.pdf
- Belasen, A., and Polachek, S. (2009). How Disasters Affect Local Labor Markets: The Effects of Hurricanes in Florida. *The Journal of Human Resources*, 44(1), 251-276.
- Belleri, A. and Marini, A. (2016). Does seismic risk affect the environmental impact of existing buildings? *Energy and Buildings*, 110, 149-158.
- Benedek, D. M., Fullerton, C., and Ursano, R. J. (2007). First responders: mental health consequences of natural and human-made disasters for public health and public safety workers. *Annual Review of Public Health*, 28, 55-68. <u>https://doi.org/10.1146/annurev.publhealth.28.021406.144037</u>.
- Berninger, A., Webber, M.P., Cohen, H.W., Gustave, J., Lee, R., Niles, J.K., Chiu, S., ZeigOwens, R., Soo, J., Kelly, K., Prezant, D.J. (2007) Trends of Elevated PTSD Risk in Firefighters Exposed to the World Trade Center Disaster: 2001-2005. *Public Health Reports*, 125, 556-566.
- Berry, S., and Davidson, K. (2016). Improving the economics of building energy code change: A review of the inputs and assumptions of economic models. *Renewable and Sustainable Energy Reviews*, 58, 157–166. <u>https://doi.org/10.1016/j.rser.2015.12.220</u>.
- Bishop, K.C., and Murphy, A.D. (2011) Estimating the Willingness to Pay to Avoid Violent Crime: A Dynamic Approach. *American Economic Review*, 101(3), 625-629.
- Black, S. E. (1999). Do better schools matter? Parental valuation of elementary education. *The Quarterly Journal of Economics*, 114(2), 577-599.

- Boardman, A. E., Greenberg, D. H., Vining, A. R., & Weimer, D. L. (2017). Cost-benefit analysis: concepts and practice (Fifth edition). Cambridge University Press, ISBN 9781108401296.
- BOMA (2018) Office Experience Exchange Reports (EER). Available at: <u>https://eer.boma.org/BOMA</u> (accessed 5 March 2019).
- Breetzke, G.D., King, M., and Fabris-Rotelli, I. (2018) The impact of the Canterbury Earthquakes on the temporal and spatial patterning of crime in Christchurch, New Zealand. *Australian & New Zealand Journal of Criminology*, 51(1), 135-156.
- Breidert, C., Hahsler, M., & Reutterer, T. (2006). A review of methods for measuring willingness-to-pay. *Innovative marketing*, 2(4), 8-32.
- Bromet, E. J., Atwoli, L., Kawakami, N., Navarro-Mateu, F., Piotrowski, P., King, A. J., Aguilar-Gaxiola, S., Alonso, J., Bunting, B., Demyttenaere, K., Florescu, S., Girolamo, G. de, Gluzman, S., Haro, J. M., Jonge, P. de, Karam, E. G., Lee, S., Kovess-Masfety, V., Medina-Mora, M. E., ... Kessler, R. C. (2017). Post-traumatic stress disorder associated with natural and human-made disasters in the World Mental Health Surveys. *Psychological Medicine*, 47(2), 227–241. https://doi.org/10.1017/S0033291716002026.
- Burton, H., Deierlien, G., Lallement, D., Lin, T. (2015) Framework for Incorporating Probabilistic Building Performance in the Assessment of Community Seismic Resilience. *Journal of Structural Engineering*, 142(8). <u>https://doi.org/10.1061/(ASCE)ST.1943-541X.0001321</u>
- Butry, D., Webb, D., Gilbert, S. and Taylor, J. (2019) The Economics of Firefighter Injuries in the United States. Technical Note (NIST TN), National Institute of Standards and Technology, Gaithersburg, MD. <u>https://doi.org/10.6028/NIST.TN.2078</u>
- Calo-Blanco, A., Kovarik, J., Mengel, F., and Romero, J.G. (2017) Natural disasters and indicators of social cohesion. PLoS ONE 12(6): e0176885. https://doi.org/10.1371/journal.pone.0176885
- Cardona, O.D., Ordaz, M.G., Yamin, L.E., Marulanda, M.C., and Barbat. A.H. (2008) Earthquake Loss Assessment for Integrated Disaster Risk Management, *Journal of Earthquake Engineering*, 12:sup2, 48-59, <u>https://doi.org/10.1080/13632460802013495</u>.
- Cardone, D., Flora, A., De Luca Picione, M., & Martoccia, A. (2019a). Estimating direct and indirect losses due to earthquake damage in residential RC buildings. Soil Dynamics and Earthquake Engineering, 126, 105801. <u>https://doi.org/10.1016/j.soildyn.2019.105801</u>.
- Cardone, D., Gesualdi, G., and Pereone, G. (2019b) Cost-Benefit analysis of alternative retrofit strategies for RC frame buildings. *J. Earthquake Engineering*, 23(2), 208-241. https://doi.org/10.1080/13632469.2017.1323041
- Cardone, D., Sullivan, T., Gesualdi, G., & Perrone, G. (2017). Simplified estimation of the expected annual loss of reinforced concrete buildings. Earthquake Engineering & Structural Dynamics, 46(12), 2009–2032. <u>https://doi.org/10.1002/eqe.2893</u>.
- Carofilis, W., Perrone, D., O'Reilly, G.J., Monteiro, R., and Filiatrault, A. (2020) Seismic retrofit of existing school buildings in Italy: Performance evaluation and loss estimation. *Engineering Structures* 225(2020): 111243. <u>https://doi.org/10.1016/j.engstruct.2020.111243</u>
- Castillo, J. C., Ahuja, A., Athey, S., Baker, A., Budish, E., Chipty, T., Glennerster, R., Kominers, S.D., Kremer, M., and Więcek, W. (2021). Market design to accelerate COVID-19 vaccine supply. *Science*, 371(6534), 1107-1109.
- Chay, K. Y., & Greenstone, M. (2005). Does air quality matter? Evidence from the housing market. *Journal of Political Economy*, 113(2), 376-424.
- Cimellaro, G., Piqué, M. (2016) Resilience of a Hospital Emergency Department under Seismic Event. Advanced in Structural Engineering, 19(5), 825-836.
- Cohen, M.A., Rust, R.R., Steen, S., Tidd, S.T. (2004). Willingness-To-Pay for Crime Control Programs. *Criminology*, 42(1), 89-109. <u>https://doi.org/10.1111/j.1745-9125.2004.tb00514.x</u>
- Comber, M.V., Poland, C., and Sinclair, M. (2012) Environmental impact seismic assessment: Application of performance-based earthquake engineering methodologies to optimize

 Comerio, M. (2006) Estimating Downtime in Loss Modeling. *Earthquake Spectra*, 22(2), 349–365.
 Comerio, M.C. (2014) Disaster Recovery and Community Renewal: Housing Approaches. <u>https://www.jstor.org/stable/26326883?seq=1</u>
 Connell, J., O'Cathain, A., and Brazier, J. (2014). Measuring quality of life in mental health: Are we asking the right questions? *Social Science & Medicine*, 120, 12–20.

https://doi.org/10.1016/j.socscimed.2014.08.026. Cook, D. (2022) Matlab codebase for quantifying building-specific functional recovery and reoccupancy based on a probabilistic performance-based earthquake engineering framework.

environmental performance. Structures Congress 2012, March 29-31, 2012, Chicago, IL.

- https://github.com/OpenPBEE/PBEE-Recovery Cook, D. and Sattar, S. (2022). The Effect of Increased Strength and Stiffness Requirements on the Functional Recovery Performance of Reinforced Concrete Special Moment Frames. 12th
 - National Conference on Earthquake Engineering, Salt Lake City, UT.
- Cook, D., Liel, A., Haselton, C., Koliou, M. (2022) A Framework for Operationalizing the Assessment of Post-Earthquake Functional Recovery of Buildings. *Earthquake Spectra:* Accepted.
- CRS (Congressional Research Service). (2011) Managing Disaster Debris: Overview of Regulatory Requirements, Agency Roles, and Selected Challenges. Congressional Research Service, Report No. RL34576. <u>https://crsreports.congress.gov/product/pdf/RL/RL34576/7</u>
- CRS (Congressional Research Service). (2021) The Disaster Relief Fund: Overview and Issues. Report No. R45484. <u>https://crsreports.congress.gov/product/pdf/R/R45484</u>
- Cutfield, M., and Ma, T. M. Q. (2015). Building owner-focussed benefit-cost analysis of seismic strengthening including market forces. Proceedings of the Tenth Pacific Conference on Earthquake Engineering Building an Earthquake-Resilient Pacific.
- D'Alpaos, C., and Bragolusi, P. (2020). The Market Price Premium for Buildings Seismic Retrofitting. Sustainability, 12(21), 8791. https://doi.org/10.3390/su12218791.
- Denver, M., Perez, J., and Aguirre, B.E. (2007) Local search and rescue teams in the United States. *Disaster Prevention and Management*, 16(4), 503-512. <u>https://doi.org/10.1108/09653560710816995</u>
- Deryugina, T. (2017) The Fiscal Cost of Hurricanes: Disaster Aid versus Social Insurance. *American Economic Journal: Economic Policy*, 9(3), 168-98.
- Di Bari, R., Belleri, A., Marini, A., Horn, R., and Gantner, J. (2020). Probabilistic Life-Cycle Assessment of Service Life Extension on Renovated Buildings under Seismic Hazard. Buildings, 10(3), 48.
- Dolgui, A., Ivanov, D., and Sokolov, B. (2018) Ripple effect in the supply chain: an analysis and recent literature. *International Journal of Production Research*, 56 (1-2), 414-430. <u>https://hal.archives-ouvertes.fr/hal-01741690</u>
- Dong, Y., and Frangopol, D.M. (2016) Performance-based seismic assessment of conventional and baseisolated steel buildings including environmental impact and resilience. *Earthquake Engineering* & Structural Dynamics 45, 739-756. <u>https://onlinelibrary.wiley.com/doi/abs/10.1002/eqe.2682</u>
- Dormady, N., Rose, A., Roa-Henriquez, A., and Morin, B. (2022). The cost-effectiveness of economic resilience. *International Journal of Production Economics*, 244, 108371. https://doi.org/10.1016/j.ijpe.2021.108371
- Doyle, J.J., and Aizer, A. (2018) Economics of Child Protection: Maltreatment, Foster Care, and Intimate Partner Violence. *Annual Review of Economics*, 10, 87-108.
- Dwyer, L., Forsyth, P., & Spurr, R. (2006). Assessing the Economic Impacts of Events: A Computable General Equilibrium Approach. *Journal of Travel Research*, 45(1), 59–66. <u>https://doi.org/10.1177/0047287506288907</u>.
- EERI (Earthquake Engineering Research Institute). (2019). Functional recovery: A conceptual framework with policy options. <u>https://www.eeri.org/images/archived/wp-content/uploads/EERI-Functional-Recovery-Conceptual-Framework-White-Paper-201912.pdf</u>

- EERI (Earthquake Engineering Research Institute). (2021) Legislative Action Center: Current Legislation and Measures. <u>https://www.eeri.org/advocacy-and-public-policy/legislative-action-center/</u>
- Eguchi, R.T., Goltz, J.D., Taylor, C.E., Chang, S.E., Flores, P.J., Johnson, L.A., Seligson, H.A., and Blais, N.C. (1998) Direct Economic Losses in the Northridge Earthquake: A Three-Year Post-Event Perspective. *Earthquake Spectra*, 14 (2), 245-264.
- EPA (2019) Planning for Natural Disaster Debris. Report No. 530-F-19-003. https://www.epa.gov/sites/default/files/2019-05/documents/final_pndd_guidance_0.pdf
- Fayers, P., and Bottomley, A. (2002). Quality of life research within the EORTC—the EORTC QLQ-C30. European Journal of Cancer, 38, 125–133. <u>https://doi.org/10.1016/S0959-8049(01)00448-</u><u>8</u>.
- Feese, C., Li, Y., and Bulleit, W.M. (2015) Assessment of Seismic Damage of Buildings and Related Environmental Impacts. J. Perform. Constr. Facil., 29(4), 04014106.
- FEMA (Federal Emergency Management Agency). (2009) BCA Reference Guide. https://www.fema.gov/sites/default/files/2020-04/fema_bca_reference-guide.pdf
- FEMA (Federal Emergency Management Agency). (2012a) Multi-hazard Loss Estimation Methodology (Hazus MH-2.1) <u>https://www.fema.gov/flood-maps/products-tools/Hazus</u>
- FEMA (Federal Emergency Management Agency). (2012b) Performance Assessment Calculation Tool (PACT). <u>https://femap58.atcouncil.org/pact</u>
- FEMA (Federal Emergency Management Agency). (2018) Seismic Performance Assessment of Buildings (P-58) <u>https://femap58.atcouncil.org/</u>
- FEMA (Federal Emergency Management Agency). (2020) Debris Removal- Reasonable Costs. <u>https://www.fema.gov/appeal/debris-removal-reasonable-costs-1</u>
- FEMA (Federal Emergency Management Agency). (2021). FEMA BCA Toolkit Version 6.0. <u>https://www.fema.gov/grants/guidance-tools/benefit-cost-analysis</u>. Accessed: 2021-10-15.
- FEMA (Federal Emergency Management Agency). (2022) Urban Search & Rescue. <u>https://www.fema.gov/emergency-managers/national-preparedness/frameworks/urban-search-rescue</u>
- Fekrazad, A. (2019). Earthquake-risk salience and housing prices: Evidence from California. Journal of Behavioral and Experimental Economics, 78, 104–113. https://doi.org/10.1016/j.socec.2019.01.001.
- Filippova, O., Rehm, M., and Dibble, C. (2017). Office market response to earthquake risk in New Zealand. *Journal of Property Investment & Finance*, 35(1), 44–57. <u>https://doi.org/10.1108/JPIF-05-2016-0026</u>.
- Filippova, O., Xiao, Y., Rehm, M., and Ingham, J. (2018). Economic effects of regulating the seismic strengthening of older buildings. *Building Research & Information*, 46(7), 711–724. <u>https://doi.org/10.1080/09613218.2017.1357318</u>.
- Fitzpatrick, A. (2020) First Responders and PTSD: A Literature Review. <u>https://www.jems.com/administration-and-leadership/first-responders-and-ptsd-a-literature-review/</u>
- Fletcher, J. D. (2010). Cost Analysis in Evaluation Studies. In P. Peterson, E. Baker, & B. McGaw (Eds.), International Encyclopedia of Education (Third Edition) (pp. 585–591). Elsevier. <u>https://doi.org/10.1016/B978-0-08-044894-7.01607-9</u>.
- Forte, F., Del Giudice, V., De Paola, P., and Del Giudice, F. P. (2021). Cultural Heritage and Seismic Disasters: Assessment Methods and Damage Types. In P. Morano, A. Oppio, P. Rosato, L. Sdino, & F. Tajani (Eds.), Appraisal and Valuation: Contemporary Issues and New Frontiers (pp. 163–175). Springer International Publishing. <u>https://doi.org/10.1007/978-3-030-49579-4_12</u>.
- Frangopol, D. M., Kong, J. S., and Gharaibeh, E. S. (2001). Reliability-Based Life-Cycle Management of Highway Bridges. *Journal of Computing in Civil Engineering*, 15(1), 27–34. <u>https://doi.org/10.1061/(ASCE)0887-3801(2001)15:1(27)</u>.

- Fung, J.F., Helgeson, J.F., Webb, D.H., O'Fallon, C.M., and Cutler H. (2021a) Does resilience yield dividends? Co-benefits of investing in increased resilience in Cedar Rapids. *Economic Systems Research*, 33:3, 336-362, <u>https://doi.org/10.1080/09535314.2020.1798359</u>.
- Fung, J.F., Sattar, S., Butry, D.T., and McCabe, S.L. (2021b) The total costs of seismic retrofits: State of the art. Earthquake Spectra, 37(4), 2991-3014. <u>https://doi.org/10.1177/87552930211009522</u>
- Fung, JF, Cook DT, Zhang Y, Johnson KJ, Sattar S (2022) Economic considerations for recovery-based design. Proceedings of the 12th National Conference in Earthquake Engineering, Earthquake Engineering Research Institute, Salt Lake City, UT.
- Fussell, J., Powers, G., and Bennetts, R. (1974) Fault Trees: A State of the Art Discussion. *IEEE Transaction on Reliability: 23(1).*
- Fussell, E., and Lowe, S. R. (2014). The impact of housing displacement on the mental health of lowincome parents after Hurricane Katrina. *Social Science & Medicine*, 113, 137–144. <u>https://doi.org/10.1016/j.socscimed.2014.05.025</u>.
- GAO (2017) Costs of Crime: Experts Report Challenges Estimating Costs and Suggest Improvements to Better Inform Policy Decisions. Report GAO-17-732. <u>https://www.gao.gov/products/gao-17-732</u>
- Ghesquiere, F., Mahul, O., and Jamin, L. (2006). Earthquake Vulnerability Reduction Program In Colombia: A Probabilistic Cost-Benefit Analysis. The World Bank. <u>https://doi.org/10.1596/1813-9450-3939</u>.
- Ghosh, S.K. (2019) "Is Seismic Design by U.S. Codes and Standards Deficient?" *Structure Magazine*. <u>https://www.structuremag.org/?p=14725</u>
- Gibson, J., Moddemeyer, S., Keithly, S., Dixon, M., and Gibson T. (2014) Seattle Unreinforced Masonry Retrofit Policy: Benefit Cost Analysis. City of Seattle Department of Planning & Development. <u>https://www.seattle.gov/Documents/Departments/SDCI/Codes/ChangesToCodes/UnreinforcedMasonry/URMBenefitsCostStudy.pdf</u>
- Gilbert, S., Butry, D., Helgeson, J. and Chapman, R. (2015), Community Resilience Economic Decision Guide for Buildings and Infrastructure Systems, Special Publication (NIST SP), National Institute of Standards and Technology, Gaithersburg, MD. <u>https://doi.org/10.6028/NIST.SP.1197</u> (Accessed August 3, 2021).
- Goettel, K.A. (2016) Benefit-Cost Analysis of the Proposed Seismic Retrofit Ordinance, City of Portland. https://www.portlandoregon.gov/pbem/article/596311
- Gonzalez, R. E., Stephens, M. T., Toma, C., & Dowdell, D. (2022). The Estimated Carbon Cost of Concrete Building Demolitions following the Canterbury Earthquake Sequence. *Earthquake* Spectra. <u>https://doi.org/10.1177/87552930221082684</u>
- Goulet, C.A., Haselton, C.B., Mitrani-Reiser, J., Beck, J.L., Deierlein, G.G., Porter, K.A., and Stewart, J.P. (2007) Evaluation of the seismic performance of a code-conforming reinforced-concrete frame building—from seismic hazard to collapse safety and economic losses. *Earthquake Engng Struct. Dyn.* 36: 1973-1997. https://doi.org/10.1002/eqe.694
- Greene, G., Paranjothy, S., and Palmer, S.R. (2015) Resilience and Vulnerability to the Psychological Harm From Flooding: The Role of Social Cohesion. *American Journal of Public Health*, 105(9), 1792-1795. <u>https://doi.org/10.2105/AJPH.2015.302709</u>
- Greenstone, M., & Gallagher, J. (2008). Does hazardous waste matter? Evidence from the housing market and the superfund program. *The Quarterly Journal of Economics*, 123(3), 951-1003.
- Guan, D., Wang, D., Hallegatte, S., Davis, S. J., Huo, J., Li, S., Bai, Y., Lei, T., Xue, Q., Coffman, D., Cheng, D., Chen, P., Liang, X., Xu, B., Lu, X., Wang, S., Hubacek, K., and Gong, P. (2020). Global supply-chain effects of COVID-19 control measures. *Nature Human Behaviour*, 4(6), 577–587. <u>https://doi.org/10.1038/s41562-020-0896-8</u>.
- Haghpanah, F., Foroughi, H., and Behrou, R. (2017) Sustainable Seismic Retrofitting of a RC Building Using Performance Based Design Approach. *Engineering Structures and Technologies* 9(3): 133-141. <u>https://doi.org/10.3846/2029882X.2017.1380539</u>

- Hallegatte, S. (2008) An adaptive regional input-output model and its application to the assessment of the economic cost of Katrina. *Risk Analysis*. 28, 779–799.
- Hallegatte, S. (2019). Disasters' impacts on supply chains. *Nature Sustainability*, 2(9), 791–792. <u>https://doi.org/10.1038/s41893-019-0380-5</u>.
- HB-Risk (Haselton Baker Risk Group, LLC). (2022) Seismic Performance Prediction Platform https://sp3risk.com/
- HCAI (2022) Preapproval Programs. https://hcai.ca.gov/construction-finance/preapproval-programs/
- Helgeson, J., P. Lavappa, and D. Webb. (2020), EDGe\$ (Economic Decision Guide Software) Online Tool, Version 1.0, National Institute of Standards and Technology, <u>https://doi.org/10.18434/M32185</u> (Accessed August 3, 2021).
- Henriet, F., Hallegatte, S., and Tabourier, L. (2012). Firm-network characteristics and economic robustness to natural disasters. *Journal of Economic Dynamics and Control*, 36(1), 150–167. <u>https://doi.org/10.1016/j.jedc.2011.10.001</u>.
- Hikichi, H., Aida, J., Tsuboya, T., Kondo, K., and Kawachi, I. (2016) Can Community Social Cohesion Prevent Posttraumatic Stress Disorder in the Aftermath of a Disaster? A Natural Experiment From the 2011 Tohoku Earthquake and Tsunami. *American Journal of Epidemiology*, 183(10), 902–910. <u>https://doi.org/10.1093/aje/kwv335</u>
- Hogg, D., Kingham, S., Wilson, T.M., and Ardagh, M. (2016) The effects of relocation and level of affectedness on mood and anxiety symptom treatments after the 2011 Christchurch earthquake. *Social Science & Medicine*, 152, 18-26. <u>https://doi.org/10.1016/j.socscimed.2016.01.025</u>.
- Horr, Y.A., Arif, M., Kaushik, A., Mazroei, A., Katafygiotou, M., and Elsarrag, E. (2016) Occupant productivity and office indoor environment quality: A review of the literature. *Building and Environment*, 105, 369-389. <u>https://doi.org/10.1016/j.buildenv.2016.06.001</u>.
- Huang, L., Frijters, P., Dalziel, K., and Clarke, P. (2018). Life satisfaction, QALYs, and the monetary value of health. *Social Science & Medicine*, 211, 131–136. <u>https://doi.org/10.1016/j.socscimed.2018.06.009</u>.
- Huang, Y. (2021). Salience of hazard disclosure and house prices: Evidence from Christchurch, New Zealand. *Regional Science and Urban Economics*, 88, 103679. <u>https://doi.org/10.1016/j.regsciurbeco.2021.103679</u>.
- ICAP (2021) USA California Cap-and-Trade Program. International Carbon Action Partnership. <u>https://icapcarbonaction.com/en/?option=com_etsmap&task=export&format=pdf&layout=list&s</u> <u>ystems%5B%5D=45</u>
- ICC (International Code Council). (2019). A National Approach to Seismic Functional Recovery for New Construction. <u>http://media.iccsafe.org/2019_MarComm/Misc/19-</u>17983_COMM_Seismic_RT_Report_FINAL_MIDrez.pdf
- ICC (International Code Council). (2020) A National Approach to Seismic Functional Recovery for New Construction: A roundtable discussion convened by the International Code Council and California Building Officials. <u>http://media.iccsafe.org/2019_MarComm/Misc/19-17983_COMM_Seismic_RT_Report_FINAL_MIDrez.pdf</u>
- ICC (International Code Council). (2021) Overview of the International Building Code. <u>https://www.iccsafe.org/products-and-services/i-codes/2018-i-codes/ibc/</u>
- IEA (2019) 2019 Global Status Report for Buildings and Construction: Towards a zero-emissions, efficient and resilient buildings and construction sector. International Energy Agency, Paris, France. <u>https://iea.blob.core.windows.net/assets/3da9daf9-ef75-4a37-b3da-</u> a09224e299dc/2019 Global Status Report for Buildings and Construction.pdf
- IFD (2021) Response Statistics. Indianapolis Fire Department (IFD), Indianapolis, IN. https://www.indy.gov/activity/fire-department-statistics

- IMDC (2020). Unveiling the Cost of Internal Displacement: 2020 Report. Internal Displacement Monitoring Centre. <u>https://www.internal-displacement.org/publications/unveiling-the-cost-of-internal-displacement-0</u>
- Inoue, H., Todo, Y. Firm-level propagation of shocks through supply-chain networks. *Nature Sustainability* 2, 841–847 (2019). <u>https://doi.org/10.1038/s41893-019-0351-x</u>.
- IRS (Internal Revenue Service). (2021) SOI Tax Stats IRS Data Book. <u>https://www.irs.gov/statistics/soi-tax-stats-irs-data-book</u>
- IWG (2016) Technical Support Document: Technical Update of the Social Cost of Carbon for Regulatory Impact Analysis - Under Executive Order 12866. Interagency Working Group on Social Cost of Greenhouse Gases, United States Government. <u>https://www.epa.gov/sites/default/files/2016-12/documents/sc co2 tsd august 2016.pdf</u>
- IWPR (2017) The Economic Cost of Intimate Partner Violence, Sexual Assault, and Stalking. Report IWPR #B367. <u>https://iwpr.org/wp-content/uploads/2020/10/B367_Economic-Impacts-of-IPV-08.14.17.pdf</u>
- Jaret, P. (2019) Hospitals on shaky ground: Meeting the challenge of seismic safety. https://www.aamc.org/news-insights/hospitals-shaky-ground-meeting-challenge-seismic-safety
- Jones, T. W., & Smith, J. D. (1982). An historical perspective of net present value and equivalent annual cost. *The Accounting Historians Journal*, 103-110.
- Joyner, M.D., and Sasani, M. (2020) Building Performance for Earthquake Resilience. *Engineering* Structures, 210, 110371. <u>https://doi.org/10.1016/j.engstruct.2020.110371</u>
- Kanya, L., Sanghera, S., Lewin, A., & Fox-Rushby, J. (2019). The criterion validity of willingness to pay methods: A systematic review and meta-analysis of the evidence. *Social Science & Medicine*, 232, 238–261. <u>https://doi.org/10.1016/j.socscimed.2019.04.015</u>.
- Kappos, A. J., and Dimitrakopoulos, E. G. (2008). Feasibility of pre-earthquake strengthening of buildings based on cost-benefit and life-cycle cost analysis, with the aid of fragility curves. *Natural Hazards*, 45(1), 33–54. <u>https://doi.org/10.1007/s11069-007-9155-9</u>.
- Katsaliaki, K., Galetsi, P., and Kumar, S. (2021) Supply chain disruptions and resilience: a major review and future research agenda. Ann Oper Res <u>https://doi.org/10.1007/s10479-020-03912-1</u>.
- Kelley, J. and Walker, M. (1959) Critical-Path Planning and Scheduling. *Proceedings of the Eastern Joint Computer Conference.*
- Kershaw, T., and Lash, D. (2013) Investigating the productivity of office workers to quantify the effectiveness of climate change adaptation measures. *Building and Environment*, 69, 35-43.
- Kircher, C., Whitman, R.V., and Holmes, W.T. (2006) HAZUS Earthquake Loss Estimation Methods. Natural Hazards Review, 7(2). <u>https://doi.org/10.1061/(ASCE)1527-6988(2006)7:2(45)</u>
- Kneifel, J. (2010). Life-cycle carbon and cost analysis of energy efficiency measures in new commercial buildings. *Energy and buildings*, 42(3), 333-340.
- Krawinkler, H., Zareian, F., Medina, R. A., & Ibarra, L. F. (2006). Decision support for conceptual performance-based design. *Earthquake Engineering & Structural Dynamics*, 35(1), 115–133. https://doi.org/10.1002/eqe.536.
- Kroll, C.A., Landis, J.D., Shen, Q., and Stryker, S. (1991) Economic Impacts of the Loma Prieta Earthquake: A Focus on Small Businesses. <u>https://escholarship.org/uc/item/05f3382m</u>
- Lachaine, J., Laurier, C., and Contandriopoulos, A.-P. (2003). Defining monetary values for quality of life improvements. *PharmacoEconomics*, 21(12), 865–874. <u>https://doi.org/10.2165/00019053-200321120-00003</u>.
- Lazrak, F., Nijkamp, P., Rietveld, P., and Rouwendal, J. (2014). The market value of cultural heritage in urban areas: An application of spatial hedonic pricing. *Journal of Geographical Systems*, 16(1), 89–114. <u>https://doi.org/10.1007/s10109-013-0188-1</u>.

- Li, Z., Li, N., Cimellaro, G. P., and Fang, D. (2019). Quantifying Hospital Resilience to Earthquakes Based on System Dynamics Modeling. ASCE International Conference on Computing in Civil Engineering 2019. <u>https://doi.org/10.1061/9780784482445.073</u>.
- Liu, M., Burns, S.A. and Wen, Y.K. (2003), Optimal seismic design of steel frame buildings based on life cycle cost considerations. *Earthquake Engng. Struct. Dyn.*, 32, 1313-1332. <u>https://doi.org/10.1002/eqe.273</u>.
- Ludin, S.M., Rohaizat, M., and Arbon, P. (2018) The association between social cohesion and community disaster resilience: A cross-sectional study. *Health and Social Care*, 27(3), 621-631.
- Manganelli, B., Vona, M., and De Paola, P. (2018). Evaluating the cost and benefits of earthquake protection of buildings. *Journal of European Real Estate Research*, 11(2), 263–278. https://doi.org/10.1108/JERER-09-2017-0029.
- Martinez-Rocamora, A., Solis-Guzman, J., and Marrero, M. (2016) LCA datasets focused on construction materials: A review. *Renewable and Sustainable Energy Reviews*, 58, 565-573.
- Meade, C., and Kulick, J. (2007). SB1953 and the Challenge of Hospital Seismic Safety in California. California Health Care Foundation, <u>https://www.eird.org/isdr-biblio/PDF/SB%201953.pdf</u>
- Mieler, M., and Mitrani-Reiser, J. (2018) Review of the State of the Art in Assessing Earthquake-Induced Loss of Functionality in Buildings. *Journal of Structural Engineering*, 144(3). https://doi.org/10.1061/(ASCE)ST.1943-541X.0001959
- Miller, T. and Hendrie, D. (2008). Substance Abuse Prevention Dollars and Cents: A Cost-Benefit Analysis. DHHS Pub. No. (SMA) 07-4298, Rockville, MD: Center for Substance Abuse Prevention, Substance Abuse and Mental Health Services Administration.
- Mitrani-Reiser, J. (2007) An Ounce of Prevention: Probabilistic Loss Estimation for Performance-Based Earthquake Engineering. Ph.D. Thesis, California Institute of Technology.
- Moehle, J., and Deierlein, G. (2004). A Framework Methodology for Performance-Based Earthquake Engineering. 13th World Conference on Earthquake Engineering. Vancouver, B.C.
- Molina-Hutt, C., Vahanvaty, T., and Kourehpaz, P. (2022) An Analytical Framework to Assess Earthquake Induced Downtime and Model Recovery of Buildings. *Earthquake Spectra*, <u>https://doi.org/10.1177/87552930211060856</u>
- Mora, M., Ordaz, M., Yamin, L., and Cardona, O. (2012) Probabilistic Cost-Benefit Ratio of Risk Mitigation Measures, 15WCEE. <u>https://www.iitk.ac.in/nicee/wcee/article/WCEE2012_1925.pdf</u>
- Morris, K. A., and Deterding, N. M. (2016). The emotional cost of distance: Geographic social network dispersion and post-traumatic stress among survivors of Hurricane Katrina. *Social Science & Medicine*, 165, 56–65. <u>https://doi.org/10.1016/j.socscimed.2016.07.034</u>.
- NAHB (2018) Estimated Costs of the 2018 ICC Code Changes for Multifamily Buildings. Report No. 3391-006_04032018. National Association of Home Builders, Upper Marlboro, MD. <u>https://www.homeinnovation.com/-/media/Files/Reports/2018-ICC-Code-Changes-for-MF-Buildings-April-2018.pdf</u>
- NEHRP Consultants Joint Venture (2013) Cost Analyses and Benefit Studies for Earthquake-Resistant Construction in Memphis, Tennessee. NIST GCR 14-917-26. https://tsapps.nist.gov/publication/get_pdf.cfm?pub_id=915569
- NIBS (2019) Natural Hazard Mitigation Saves: 2019 Report. <u>https://www.nibs.org/projects/natural-hazard-mitigation-saves-2019-report</u>
- NIST (National Institute of Standards and Technology). (2018) Research Needs to Support Immediate Occupancy Building Performance Objective Following Natural Hazard Events. Sattar, S., McAllister, T., Johnson, K., Clavin, C., Segura, C., McCabe, S., Fung, J., Abrahams, L., Sylak-Glassman, E., Levitan, M., Harrison, K., Harris, J. National Institute of Standards and Technology. <u>https://nvlpubs.nist.gov/nistpubs/SpecialPublications/NIST.SP.1224.pdf</u>
- NIST (National Institute of Standards and Technology). (2021) "NIST SP-1254: Recommended Options for Improving the Built Environment for Post-Earthquake Reoccupancy and Functional Recovery Time." <u>https://doi.org/10.6028/NIST.SP.1254</u>
- Nuti, C., and Vanzi, I. (2003). To retrofit or not to retrofit? *Engineering Structures*, 25(6), 701–711. https://doi.org/10.1016/S0141-0296(02)00190-6.
- Odabaşi, Ö., Kohrangi, M., Fagà, E., and Bazzurro, P. (2020). Consequence Functions for Seismic Risk Assessment: A review of consequence modelling state-of-practice for ERM-CH – Module D [Report]. ETH Zurich. <u>https://doi.org/10.3929/ethz-b-000416778</u>.
- Okuyama, Y. (2007). Economic Modeling for Disaster Impact Analysis: Past, Present, and Future. *Economic Systems Research*, 19(2), 115–124. <u>https://doi.org/10.1080/09535310701328435</u>.
- Owen, S. and Noy, I. (2017) The Unfortunate Regressivity of Public Natural Hazard Insurance: A Quantitative Analysis of a New Zealand Case. CESifo Working Paper Series No. 6540. https://ssrn.com/abstract=3005165
- Paul, N., Lee, J., Mieler, M., and Almifti, I. (2018) Improving Estimates of Earthquake-Induced Downtime in Individual Buildings Using the REDi Methodology. Structures Congress, April 19– 21, 2018, Fort Worth, Texas. <u>https://doi.org/10.1061/9780784481349.008</u>
- Paxton, B., Elwood, K.J., and Ingham, J.M. (2017) Empirical Damage Relationships and Benefit-Cost Analysis for the Seismic Retrofit of URM Buildings. *Earthquake Spectra* 33(3): 1053-1074. <u>https://doi.org/10.1193/091816eqs153m</u>
- Pearce, D., Atkinson, G., & Mourato, S. (2006). Cost-benefit analysis and the environment: recent developments. Organisation for Economic Co-operation and development, Paris, France. ISBN 9264010041.
- PEER (Pacific Earthquake Engineering Research Institute). (2017) Tall Building Initiative: Guidelines for Performance-Based Seismic Design of Tall Buildings. <u>https://peer.berkeley.edu/sites/default/files/tbi-working-group-led-by-co-chairs-ron-ha-burger-and-jack-moehle_10.9.17_withlinks.pdf</u>
- Petak, W.J., and Elahi, S. (2001) The Northridge Earthquake, USA and its Economic and Social Impacts. EuroConference on Global Change and Catastrophe Risk Management Earthquake Risks in Europe, IIASA, Laxenburg, Austria, July 6-9, 2000. <u>https://abag.ca.gov/sites/default/files/the_northridge_earthquake_and_its_economic_and_social_i</u> <u>mpacts.pdf</u>
- Peterson, J., and Small, M. J. (2012). Methodology for benefit–cost analysis of seismic codes. *Natural Hazards*, 63(2), 1039–1053. <u>https://doi.org/10.1007/s11069-012-0204-7</u>.
- Pew (2018) What We Don't Know About State Spending on Natural Disasters Could Cost Us: Data limitations, their implications for policymaking, and strategies for improvement. <u>https://www.pewtrusts.org/en/research-and-analysis/reports/2018/06/19/what-we-dont-know-about-state-spending-on-natural-disasters-could-cost-us</u>
- Phillip, A. (2015) White people in New Orleans say they're better off after Katrina. Black people don't. <u>https://www.washingtonpost.com/news/post-nation/wp/2015/08/24/white-people-in-new-orleans-say-theyre-better-off-after-katrina-black-people-dont/</u>
- Pope, D.G., and Pope, J.C. (2012) Crime and property values: Evidence from the 1990s crime drop. *Regional Science and Urban Economics*, 42(1-2), 177-188.
- Porter, K. (2003) An Overview of PEER's Performance-Based Earthquake Engineering Methodology. Ninth International Conference on Applications of Statistics and Probability in Civil Engineering (ICASP9) July 6-9, 2003, San Francisco.
- Porter, K. A., Beck, J. L., & Shaikhutdinov, R. (2004). Simplified Estimation of Economic Seismic Risk for Buildings. Earthquake Spectra, 20(4), 1239–1263. <u>https://doi.org/10.1193/1.1809129</u>.

- Porter, K. and Ramer, K. (2012) Estimating Earthquake-Induced Failure Probability and Downtime of Critical Facilities. *Journal of Business Continuity and Emergency Planning: 5(4)*. https://www.sparisk.com/pubs/Porter-2012-JBCEP-Downtime.pdf
- Porter, K.A. (2016) Not safe enough: the case for resilient seismic design. 2016 SEAOC Convention, October 12-15, 2016, Maui HI. <u>https://www.sparisk.com/pubs/Porter-2016-SEAOC-Resilience.pdf</u>
- Porter, K.A., Scawthorn, C.R., and Beck, J.L. (2006) Cost-Effectiveness of Stronger Woodframe Buildings. *Earthquake Spectra* 22(1), 239-266. <u>https://doi.org/10.1193/1.2162567</u>
- Potter, S.H., Becker, J.S., Johnston, D.M., Rossiter, K.P. (2015) An overview of the impacts of the 2010-2011 Canterbury earthquakes. https://www.sciencedirect.com/science/article/pii/S221242091500031X
- Powell, F.I., Harding, A.J.M., and Beetham, J.M. (2015) Developing an evidence base of the outcomes of retrofitting commercial properties. Proceedings of the 2015 NZSEE Conference, 10 – 12 April 2015, Rotorua, New Zealand.
- Prelog, A.J. (2016) Modeling the Relationship between Natural Disasters and Crime in the United States. Natural Hazards Review, 17(1), 04015011. <u>https://doi.org/10.1061/(ASCE)NH.1527-6996.0000190</u>
- Preston, B.L., LaTourretle, T., Broyles, J.R., Briggs, R.J., Catt, D., Nelson, C., Ringel, J.S., Waxman, D.A. (2019) Updating the Costs of Compliance for California's Hospital Seismic Safety Standards. RAND Corporation, Santa Monica, CA. https://www.rand.org/pubs/research_reports/RR3059.html.
- Ramirez, C. M., Liel, A. B., Mitrani-Reiser, J., Haselton, C. B., Spear, A. D., Steiner, J., Deierlein, G. G., & Miranda, E. (2012). Expected earthquake damage and repair costs in reinforced concrete frame buildings. *Earthquake Engineering & Structural Dynamics*, 41(11), 1455–1475. <u>https://doi.org/10.1002/eqe.2216</u>.
- REDi (2013) REDi Rating System. https://www.arup.com/perspectives/publications/research/section/redi-rating-system
- RGGI (2021) The Regional Greenhouse Gas Initiative: An Initiative of Eastern States of the US. https://www.rggi.org/sites/default/files/Uploads/Fact%20Sheets/RGGI_101_Factsheet.pdf
- RS Means (2021). Available at https://rsmeansonline.com. Retrieved: 2021-05-01.
- Ryu, H., Karaca, E., Luco, N., and Milburn, T. (2010) Regional risk assessment for seismic design alternatives - the case of Memphis, Tennessee. Proceedings of the 9th US National and 10th Canadian Conference on Earthquake Engineering, July25-29, 2010, Toronto, Ontario, Canada. Paper No. 966.
- Sarmiento, C. (2007) The impact of flood hazards on local employment. *Applied Economics Letters*, 14:15, 1123-1126, DOI: 10.1080/13504850600606026
- Seppanen, O., Fisk, W.J., Faulkner, D. (2004) Control of temperature for health and productivity in offices. <u>https://escholarship.org/content/qt39s1m92c/qt39s1m92c.pdf</u>
- Shigemoto, Y., and Kawachi, I. (2020) Social cohesion and quality of life among survivors of a natural disaster. *Quality of Life Research*, 29, 3191–3200.
- Sidney, J. A., Jones, A., Coberley, C., Pope, J. E., and Wells, A. (2017). The well-being valuation model: A method for monetizing the nonmarket good of individual well-being. *Health Services and Outcomes Research Methodology*, 17(1), 84–100. <u>https://doi.org/10.1007/s10742-016-0161-9</u>.
- Simmons, K.M., Czajkowski, J., and Done, J. (2017) Economic Effectiveness of Implementing a Statewide Building Code: The Case of Florida. *Land Economics* 94(2): 155-174. <u>http://dx.doi.org/10.2139/ssrn.2963244</u>
- Simmons, K.M., kovacs, P., and Kopp, G.A. (2015) Tornado Damage Mitigation: Benefit–Cost Analysis of Enhanced Building Codes in Oklahoma. *Weather, Climate, and Society* 7(2): 169-178. <u>https://doi.org/10.1175/WCAS-D-14-00032.1</u>

- Singh, R. (2019). Seismic risk and house prices: Evidence from earthquake fault zoning. *Regional Science* and Urban Economics, 75, 187–209. <u>https://doi.org/10.1016/j.regsciurbeco.2019.02.001</u>.
- Smyth, A.W., Altay, G., Deodatis, G., Erdik, M., Franco, G., Gulkan, P., Kunreuther, H., Lus, H., Mete, E., Seeber, N., and Yuzugullu, O. (2004) Probabilistic Benefit-Cost Analysis for Earthquake Damage Mitigation: Evaluating Measures for Apartment Houses in Turkey. *Earthquake Spectra* 20(1): 171-203. <u>https://doi.org/10.1193/1.1649937</u>
- Stein, J., and Van Dam, A. (2019) Taxpayer spending on U.S. disaster fund explodes amid climate change, population trends. <u>https://www.washingtonpost.com/us-policy/2019/04/22/taxpayer-spending-us-disaster-fund-explodes-amid-climate-change-population-trends/</u>
- Stevens, T. H., Belkner, R., Dennis, D., Kittredge, D., & Willis, C. (2000). Comparison of contingent valuation and conjoint analysis in ecosystem management. *Ecological Economics*, 32(1), 63–74. <u>https://doi.org/10.1016/S0921-8009(99)00071-3</u>.
- Sunstein, C. (2007). Willingness to Pay versus Welfare. Law & Economics Working Papers. https://chicagounbound.uchicago.edu/law_and_economics/627.
- Sunstein, C. (2013). The value of a statistical life: Some clarifications and puzzles. *Journal of Benefit-Cost Analysis*, 4(2), 237-261. <u>https://doi.org/10.1515/jbca-2013-0019</u>.
- Taks, Marijke; Kesenne, Stefan; Chalip, Laurence; and Green, Christine B. (2011). Economic Impact Analysis Versus Cost Benefit Analysis: The Case of a Medium-Sized Sport Event. International Journal of Sport Finance, 6 (3), 187-203. <u>https://scholar.uwindsor.ca/humankineticspub/20</u>.
- Taylor, A. (2014) The Northridge Earthquake: 20 Years Ago Today. <u>https://www.theatlantic.com/photo/2014/01/the-northridge-earthquake-20-years-ago-today/100664/</u>
- Terzic, V., and Kolozvari, K. (2020) Evaluation of Post-Earthquake Functional Recovery for a 42-Story Building Located in Los Angeles. Proceedings of the 2020 Los Angeles Tall Buildings Structural Design Council Conference, 93-113.
- Terzic, V., and Villanueva, P. (2021) Method for probabilistic evaluation of post-earthquake functionality of building systems. *Engineering Structures*, 241.
- Thomas, D., Butry, D., Gilbert, S., Webb, D. and Fung, J. (2017), The Costs and Losses of Wildfires, Special Publication (NIST SP), National Institute of Standards and Technology, Gaithersburg, MD, [online], <u>https://doi.org/10.6028/NIST.SP.1215</u>.
- Thomas, D., and Helgeson, J. (2021) The effect of natural/human-made hazards on business establishments and their supply chains. *International Journal of Disaster Risk Reduction*, 59, 102257.
- Thornley, L., Ball, J., Signal, L., Lawson-Te Aho, K., and Rawson, E. (2015) Building community resilience: learning from the Canterbury earthquakes. Kotuitui: New Zealand Journal of Social Sciences Online, 10(1), 23-35. <u>https://doi.org/10.1080/1177083X.2014.934846</u>
- Timar, L., Grimes, A., and Fabling, R. (2018a). Before a Fall: Impacts of Earthquake Regulation on Commercial Buildings. *Economics of Disasters and Climate Change*, 2(1), 73–90. <u>https://doi.org/10.1007/s41885-017-0019-9</u>.
- Timar, L., Grimes, A., and Fabling, R. (2018b). That sinking feeling: The changing price of urban disaster risk following an earthquake. *International Journal of Disaster Risk Reduction*, 31, 1326–1336. <u>https://doi.org/10.1016/j.ijdrr.2018.01.026</u>.
- TMS (The Masonry Society). (2021) TMS 402/602-16 Building Code Requirements and Specification for Masonry Structures. <u>https://masonrysociety.org/product/tms-402-602-2016/</u>
- Tokas, C. (2011). Nonstructural Components and Systems Designing Hospitals for Post Earthquake Functionality. Proceedings of the ASCE & Architectural Engineering Institute Conference, Oakland, CA.
- Townshend, I., Awosoga, O., Kulig, J., Fan, H. (2015) Social cohesion and resilience across communities that have experienced a disaster. *Nat Hazards*, 76, 913–938.

- Tulchinsky, T. H., & Varavikova, E. A. (2014). Chapter 11 Measuring Costs: The Economics of Health. In T. H. Tulchinsky & E. A. Varavikova (Eds.), The New Public Health (Third Edition) (pp. 575– 611). Academic Press. <u>https://doi.org/10.1016/B978-0-12-415766-8.00011-2</u>.
- Turner, F. (2004) Seventy Years of the Riley Act and its Effect on California's Building Stock. 13th World Conference on Earthquake Engineering, August 1-6, 2004, Vancouver, B.C., Canada. Paper No. 313. <u>https://www.iitk.ac.in/nicee/wcee/article/13_313.pdf</u>
- Uscher-Pines, L. (2009). Health effects of relocation following disaster: A systematic review of the literature. *Disasters*, 33(1), 1–22. <u>https://doi.org/10.1111/j.1467-7717.2008.01059.x</u>.
- USGS (2021) HayWired Scenario. <u>https://www.usgs.gov/natural-hazards/science-application-risk-</u>reduction/science/haywired-scenario?qt-science center objects=0#qt-science center objects
- USRC (United States Resilience Council). (2015) USRC Building Rating System for Earthquake Hazards: Implementation Manual. <u>https://www.usrc.org/</u>
- Wanner GK, Loyd JW. (2021) EMS Care Teams In Disaster Response. StatPearls Publishing. Available from: <u>https://www.ncbi.nlm.nih.gov/books/NBK482333/</u>
- Weisbrod, G., & Weisbrod, B. (1997). Measuring economic impacts of projects and programs. Economic Development Research Group, 10, 1-11.
- Weisbrod, G., Mulley, C., & Hensher, D. (2016). Recognising the complementary contributions of cost benefit analysis and economic impact analysis to an understanding of the worth of public transport investment: A case study of bus rapid transit in Sydney, Australia. *Research in Transportation Economics*, 59, 450-461. <u>https://doi.org/10.1016/j.retrec.2016.06.007</u>.
- Welch, D. P., Sullivan, T. J., & Calvi, G. M. (2014). Developing Direct Displacement-Based Procedures for Simplified Loss Assessment in Performance-Based Earthquake Engineering. *Journal of Earthquake Engineering*, 18(2), 290–322. <u>https://doi.org/10.1080/13632469.2013.851046</u>.
- Wen, Y. K., & Kang, Y. J. (2001). Minimum building life-cycle cost design criteria. I: Methodology. Journal of Structural Engineering, 127(3), 330-337.
- Whitehead, J.C. (2003) One Million Dollars per Mile? The Opportunity Costs of Hurricane Evacuation. Ocean and Coastal Management. 46(11-12), 1069-1083. https://libres.uncg.edu/ir/asu/f/Whitehead John 2003 One Million Dollars.pdf
- Whitehead, S. J., and Ali, S. (2010). Health outcomes in economic evaluation: The QALY and utilities. *British Medical Bulletin*, 96(1), 5–21. <u>https://doi.org/10.1093/bmb/ldq033</u>.
- Wisnivesky, J.P., Teitelbaum, S.L., Todd, A.C., Boffetta, P., Crane, M., Crowley, L., de la Hoz, R.E., Dellenbaugh, C., Harrison, D., Herbert, R., Kim, J., Jeon, Y., Kaplan, J., Katz, C., Levin, S., Luft, B., Markowitz, S., Moline, J.M., Ozbay, F., Pietrzak, R.H., Shapiro, M. Sharma, V., Skloot, G., Southwick, S., Stevenson, L.A., Udasin, I., Wallenstein, S., Landrigan, P.J. (2011) Persistence of Multiple Illnesses in World Trade Center Rescue and Recovery Workers: A Cohort Study. *Lancet*, 378, 888-897.
- Yabe, T., Tsubouchi, K., Fujiwara, N., Sekimoto, Y., and Ukkusuri, S.V. (2020) Understanding postdisaster population recovery patterns. J. R. Soc. Interface., 17, 20190532. <u>http://doi.org/10.1098/rsif.2019.0532</u>
- Yashinsky, M. (1998) The Loma Prieta, California, Earthquake of October 17, 1989--Highway Systems. https://pubs.usgs.gov/pp/pp1552/pp1552b/pp1552b.pdf
- Ye, L., and Abe, M. (2012) The impacts of natural disasters on global supply chains. ARTNeT Working Paper Series, No. 115, Asia-Pacific Research and Training Network on Trade (ARTNeT), Bangkok. https://www.econstor.eu/bitstream/10419/64267/1/717874087.pdf
- Zahran S, Shelley O'Connor T, Peek L, Brody SD. (2009) Natural disasters and social order: modeling crime outcomes in Florida. *Int J Mass Emerg Disasters*, 27(1), 26–52.
- Zerbe, R. O., & Bellas, A. S. (2006). A primer for benefit-cost analysis. Edward Elgar Publishing, ISBN 9781843768975.

Zhang, Y., Fung, J.F., Johnson, K., and Sattar, S. (2021) Review of Seismic Risk Mitigation Policies in Earthquake-Prone Countries: Lessons for Earthquake Resilience in the United States. *Journal of Earthquake Engineering*, 1-18. <u>https://doi.org/10.1080/13632469.2021.1911889</u>

Zhang, Y., Fung, J.F., Johnson, K.J., Sattar S., and Cook, D. (2022) Benefit-cost analysis for earthquake resilient building design: The state of the art and future research needs. Unpublished manuscript.

Occupancy	Definition	Examples
group		
Ι	Buildings and structures that represent low risk to human	Barns and storage shelters
	life in the event of failure	
II	All buildings and structures except those listed in Risk	Residential, commercial
	Categories I, III and IV	and industrial buildings
III	• Buildings and other structures, the failure of	• Theaters, lecture
	which could pose a substantial risk to human life	halls, and dining
	• Buildings and other structures, not included in	halls
	Risk Category IV, with potential to cause a	• Grade schools,
	substantial economic impact and/or mass	prisons, and small
	disruption of day-to-day civilian life in the event	healthcare facilities
	of failure	• Power-generating
	• Building and other structures not included in Risk	stations, water
	Category IV, containing toxic or explosive	treatment, and
	substances where the quantity of material exceeds	sewage treatment
	a threshold quantity established by the authority	plants
	having jurisdiction and is sufficient to pose a	
	threat to the public if released	

Table A-1. Risk category of buildings and other structures	for earthquake loads	(ASCE 7-16).
--	----------------------	--------------

IV	• Building and other structures designed as essential	Hospitals, police stations,
	facilities	emergency control centers
	• Building and other structures, the failure of which	
	could pose a substantial hazard to the community	
	• Building and other structures containing sufficient	
	quantities of highly toxic substances where the	
	quantity of material exceeds a threshold quantity	
	established by the authority having jurisdiction	
	and is sufficient to pose a threat to the public if	
	released	
	• Building and other structures required to maintain	
	the functionality of other Risk Category IV	
	structures	

Table A-2. Seismic design category (ASCE 7-16).

Design category	Definition
A	Buildings in areas where expected ground shaking is minor
В	Buildings of Occupancy Groups I, II and III where expected ground shaking is moderate
С	Buildings of Occupancy Groups IV where expected ground shaking is moderate and buildings of Occupancy Groups I, II, and III where more severe ground shaking will occur
D	Buildings in areas expected to experience severe and destructive ground shaking but not located close to a major fault
E	Buildings of Occupancy Groups I, II and III in areas near major active faults
F	Buildings of Occupancy Groups IV in areas near major active faults

Appendix B. Methods/Tools for estimating costs

There are many tools designed to assist in cost estimating. Below lists the tools that are readily available to any users.

- Craftsman National Cost Estimator, <u>http://www.craftsmansitelicense.com</u>
- Standard Estimating Practice edited by American Society of Professional Estimators (ASPE), <u>https://www.aspenational.org/page/SEP</u>
- Autodesk Construction Cloud, <u>https://go.construction.autodesk.com/demo-request/</u>
- Corecon Technologies Cost Database, <u>https://www.corecon.com/estimating-cost-databases</u>
- EG Sigma Software, <u>https://sigmaestimates.com/products/cost-database-construction/</u>
- Design Cost Data (DCD), <u>https://www.dcd.com/articles/category/estimating</u>
- HomeAdvisor seismic retrofit costs by zip codes, <u>https://www.homeadvisor.com/cost/environmental-safety/earthquake-retrofit-a-home/</u>

Yet novel methods are needed for estimating construction costs and retrofit costs, such as using machine learning methods trained on cost data (Fung et al. 2017, 2018a, 2018b, 2019, 2020). Note that a database of retrofit costs for functional recovery is not available, except maybe for Risk Category IV, but this is not generalizable to other buildings. One potential direction for future research is to collect engineering consulting estimates for different building prototypes, but this faces the same problem as to how to generalize. Another potential direction for future research is to develop an analytical model that minimizes "annual equivalent cost," as discussed in Nuti and Vanzi (2003).